ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hubrich, Kirstin; Waggoner, Daniel F.

Working Paper The transmission of financial shocks and leverage of financial institutions: An endogenous regime-switching framework

Working Paper, No. 2022-5

Provided in Cooperation with: Federal Reserve Bank of Atlanta

Suggested Citation: Hubrich, Kirstin; Waggoner, Daniel F. (2022) : The transmission of financial shocks and leverage of financial institutions: An endogenous regime-switching framework, Working Paper, No. 2022-5, Federal Reserve Bank of Atlanta, Atlanta, GA, https://doi.org/10.29338/wp2022-05

This Version is available at: https://hdl.handle.net/10419/270448

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

The Transmission of Financial Shocks and Leverage of Financial Institutions: An Endogenous Regime-Switching Framework

Kirstin Hubrich and Daniel Waggoner

Working Paper 2022-5 June 2022

Abstract: We conduct a novel empirical analysis of the role of leverage of financial institutions for the transmission of financial shocks to the macroeconomy. For that purpose, we develop an endogenous regime-switching structural vector autoregressive model with time-varying transition probabilities that depend on the state of the economy. We propose new identification techniques for regime switching models.

Recently developed theoretical models emphasize the role of bank balance sheets for the buildup of financial instabilities and the amplification of financial shocks. We build a market-based measure of leverage of financial institutions employing institution-level data and find empirical evidence that real effects of financial shocks are amplified by the leverage of financial institutions in the financial constraint regime. We also find evidence for a role of heterogeneity of financial institutions including depository financial institutions, globally systemically important banks, and selected nonbank financial institutions. Our results suggest that the leverage ratio is a useful indicator from a policy perspective.

JEL classification: C11, C32, C53, C55, E44, G21

Key words: regime switching, time-varying transition probabilities, financial shocks, leverage, bank and nonbank financial institutions, heterogeneity

https://doi.org/10.29338/wp2022-05

The views expressed here are those of the authors and not necessarily those of the Federal Reserve Bank of Atlanta or the Federal Reserve System. Any remaining errors are the authors' responsibility. The authors thank Francesco Ferrante, Robert Tetlow, and Tao Zha for valuable discussions and Jose Berrospide, Richard Clarida, Jurgen Doornik, Nathan Foley-Fisher, Jesper Linde, Helmut Luetkepohl, Sophokles Mavroides, Teodora Paligorova, Lubomir Petrasek, Chris Sims, Gustavo Suarez, Skander van den Heuvel, and Min Wei. They also thank seminar participants at Oxford University, the Federal Reserve Board, the Federal Reserve Bank of Atlanta, the Federal Reserve System, the International Monetary Fund, the European Economic Association, the International Association of Applied Econometrics, NBER-NSF conferences, and the conference "Innovations in Time Series Econometrics" in Berlin for useful comments. They gratefully acknowledge valuable research assistance by Kevin Starnes, Rebecca John, Jack McCoy, and Hongyi Wu. This paper was previously circulated under the title "The Transmission of Financial Shocks and Leverage of Banks: An Endogenous Regime Switching Framework."

Please address questions regarding content to Kirstin Hubrich, Federal Reserve Board, kirstin.hubrich@frb.gov, and Daniel Waggoner, Federal Reserve Bank of Atlanta, daniel.f.waggoner@atl.frb.org.

Federal Reserve Bank of Atlanta working papers, including revised versions, are available on the Atlanta Fed's website at www.frbatlanta.org. Click "Publications" and then "Working Papers." To receive e-mail notifications about new papers, use frbatlanta.org/forms/subscribe.

1 Introduction

Since the Global Financial Crisis (GFC) progress has been made in understanding the interactions of financial constraints, financial market instabilities and the macroeconomy and incorporating those in standard macreconomic models, but further work is needed.

There is a growing literature incorporating more elaborate representations of financial markets into structural macroeconomic models, in particular bank balance sheets and leverage of financial institutions. The number of contributions incorporating nonlinearities in that context is, however, still limited.

Our contributions in this paper are to propose an endogenous switching model framework and new identification techniques as well as providing empirical evidence of the relevance of (market-based) leverage of financial institutions for the transmission of shocks to the macroeconomy and the role of heterogeneity of financial institutions including banks and nonbanks in this context.

The GFC illustrated how a financial crisis might arise via the supply side of financial markets. In other situations, imbalances in the financial sector have amplified macroeconomic fluctuations and downturns, for instance during the dot com bubble. While the origin of the current crisis lies in a pandemic - different from the GFC -, financial factors have played an important role at the onset of the current crisis. Financial institutions entered the pandemic crisis better capitalized than the GFC and financial markets calmed due to an early and powerful central bank response. The increasing role of non-banks suggests a particular focus on the heterogeneity of different financial institutions.¹

With this paper we contribute to the literature in a number of ways with methodological developments as well as the empirical analysis. First, we provide methodological contributions. We propose a regime switching vector autoregressive (RS-VAR) model with time-varying transition probabilities. We show how the Markov-switching structural vector autoregression model framework, proposed in Sims and Zha (2006) and Sims et al. (2008), and employed in Hubrich and Tetlow (2015) can be extended in several dimensions. More precisely, we

¹Financial institutions entered the crisis better capitalized due to improved regulations following the GFC, in contrast to non-financial firms entered the current crisis with a high level of indebtedness. However, solvency problems of nonfinancial firms will have consequences both for the financial system as well as for the macroeconomy. Therefore, it is important also in the context of the current crisis to understand better the role of banks for the transmission of shocks.

extend previous literature on Markov-Switching models in two important dimensions: 1. We allow for time-varying probabilities in RS-VAR models. Thereby we go beyond the singleequation models suggested in the literature. In regime switching models with time-varying probabilities - sometimes referred to as "endogenous switching" models - the transition probability of being in one regime in the next period, given that we are in a particular regime in this period, can vary over time. Therefore, the transition matrix can depend on the state of the economy. 2. We propose new identification techniques for RS-VAR models, allowing a range of general, non-recursive (over-)identification schemes of the structural shocks. Besides nonrecursive zero restrictions, these also include sign restrictions and narrative sign restrictions, thereby bringing the approaches suggested in Antolin-Diaz and Rubio-Ramirez (2018) and Arias, Rubio-Ramirez and Waggoner (2018) to the class of regime switching models. We also allow for different identification schemes in different regimes. One additional contribution is our systematic discussion of identification issues in regime switching models. We employ the recently developed Dynamic Striated Metropolis-Hastings sampler for high-dimensional models to estimate the posterior distribution of the model. This new framework allows us to address the economic questions raised above.

Second, employing this new modelling framework we provide a novel empirical analysis of the role of leverage of financial institutions for the transmission of financial shocks to the macroeconomy. The motivation for our focus on leverage is threefold: First, recent literature on structural macroeconomic models emphasizes the role of bank balance sheets for the build-up of financial instabilities and the amplification of economic downturns. Second, leverage encompasses the entire balance sheet of the financial institution and therefore is a broad indicator for signaling financial vulnerabilities. Third, the leverage ratio is a regulatory tool complementary to the (risk-weighted) capital ratio.

We build a market-based measure of leverage of financial institutions, building on Adrian and Brunnermeier $(2016)^2$. We employ financial institution level data to construct a monthly measure of leverage as book assets over market equity. Two arguments suggest a focus on market leverage: First, Market leverage developments can signal a situation where financial institutions need to deleverage quickly, for instance if debt is used to finance asset growth as

²See also Paul (2020)

for broker-dealers (see Adrian and Shin (2014)) and if financial institutions rely primarily on short-term funding (see e.g. Adrian et al. (2011) and also the related literature on maturity transformation). More generally, we argue that monitoring market leverage is relevant for a range of reasons including the effect on financial institutions for coping with liquidity shocks and on funding costs (there is evidence that the cost advantage is larger for banks that are thinly capitalized) relevant from a financial stability perspective related to the liquidity regulations introduced in Basel III. Second, market values of equity are more informative about financial institutions' losses compared to book values. Book equity values might not be a timely predictor of bank health. Since book values incorporate information on losses with a delay, financial institutions have time to adjust their book leverage in order to avoid hitting the regulatory limit.³ Financial institutions (banks and non-bank financial institutions) might be more fragile than their book leverage levels make them appear. Furthermore, market capitalization of a bank is a reflection of the market value of the equity holders' stake, and hence an assessment by market participants of the creditworthiness of the bank as a borrower. Low market-to-book ratios suggest that the assessment of market participants is that banks are more leveraged than their books suggest (see also Adrian et al. (2018)). We highlight the role of the financial fragility implied by market leverage for the transmission of financial shocks in our empirical model, that has been pointed to in a model estimated to match four facts about banks' leverage dynamics (see Begenau et al. (2021)).

We contribute to the empirical literature by, first, presenting empirical evidence for the role of (market) leverage for the amplification of the transmission of financial shocks and the implications for the real economy; second, providing empirical evidence for a different transmission of financial shocks in different regimes; and third, providing evidence of a role of the heterogeneity of financial institutions' leverage for the detrimental real effects of deleveraging of financial institutions and the implications for the probability of persistence of the financial constraint regime. In particular, we consider Depository Financial Institutions, Globally Systemically Important Banks (GSIBs) as well as selected Non-Bank Financial Institutions.

The paper is structured as follows. Section 2 discusses the related theoretical and em-

³So far the information content of market equity about book losses has been mostly highlighted in the accounting literature, indicating that banks have flexibility in accounting for losses, consistent with evidence in Blattner et al. (2022). This flexibility in accounting for losses can be even more prominent for non-bank financial institutions that are part of our analysis.

pirical literature, thereby motivating the models estimated in this paper. Section 3 presents our new methodological proposal, outlines the estimation and evaluation of the model and discusses the identification issues that arise for RS-VAR models. Section 4 contains the economic motivation, the data and model specification and the empirical results. Section 5 concludes.

2 Related economic literature and contribution of this paper

We complement previous empirical studies on financial constraints and economic dynamics by providing empirical evidence on the role of leverage of financial institutions for the transmission of financial shocks to the macroeconomy, and in particular market-based leverage. We develop a new regime switching SVAR model framework for this analysis that is motivated by recent theoretical model developments.

Endogenous financial instabilities: Theoretical Models

The theoretical literature has made progress recently in incorporating financial instability and associated nonlinearities into macroeconomic models. Structural models such as Kiyotaki and Moore (1997) as well as Brunnermeier and Sannikov (2014) illustrate how systemic risk might arise endogenously, determined by the choices of the model's decision makers. In Kiyotaki and Moore (1997) collateral constraints play a key role for the propagation and amplification of shocks while in Brunnermeier and Sannikov (2014) the reduction in the volatility of output and asset prices leads to increased leverage of financial institutions.⁴

Bank balance sheets and leverage of financial institutions

More recently, a number of authors introduced a more sophisticated financial sector into an otherwise standard macroeconomic model. Financial intermediaries' balance sheets have implications for the institutions' access to funds and liquidity that affect their lending activities and thereby economic activity. A fall in the value of a bank's tradable assets and a decline in loan quality can adversely affect the bank's capital. The fall in asset prices will affect lending activity via the collateral channel. Banks have been found to limit their deposit taking

⁴Other contributions that incorporate financial instabilities and nonlinearities include Mendoza (2010) and He and Krishnamurthy (2019) and Boissay et al. (2016).

in response to a decline in net worth Gertler and Kiyotaki (2010, 2015); Gertler and Karadi (2013).

The importance of the bank capital channel will also depend on the extent to which nonbank financial institutions can substitute lending and liquidity provisions by banks (see e.g. Durdu and Zhong (2019)).

Leverage of financial institutions is an important characteristic in the presence of large and abrupt asset price movements (see e.g. Gertler and Gilchrist (2018) on the role of leverage, and Adrian and Brunnermeier (2016) and Paul (2020) for more details on the mechanisms). The leverage ratio of a bank is the ratio of total assets to shareholder equity. Bank leverage is an indicator for external financing opportunities by banks and for risk-taking by banks. It is a cyclical indicator and can amplify the transmission of shocks.

Asset prices affect the balance sheet and thereby affect leverage both in an accounting sense as well as via resulting changes in the agents' behavior (see Paul (2020)). Financial vulnerabilities build up in boom times, when banks enlarge their balance sheets and increase their leverage, relying more on debt as opposed to equity. In the lead-up to the GFC and Great Recession there was a pronounced rise in leverage in the banking sector. During the GFC stock prices fell dramatically, increasing the already high leverage of banks even further. At the same time, banks' market value of assets in terms of share holdings was also shrinking, reducing the access of banks to external finance (see e.g. Ferrante (2019)).⁵ At some point, banks had to sharply reduce the provision of loans to households and firms to obtain liquidity to avoid a bank run and insolvency (see e.g. Gertler et al. (2016) andGertler et al. (2020)). Consequently, banks were more constrained in raising funds and therefore were providing fewer/lower volume loans.⁶ Additionally, when banks realized that this shock was not temporary, they deleveraged given a reduced risk tolerance, adding further constraints. ⁷ The discussion around Basel III and its implementation has aimed to address capital and leverage

⁵Ferrante (2019), building on the framework of Gertler and Karadi (2011), extends a standard New Keynesian model to include a rich financial system in which financially constrained banks lend to firms and homeowners via defaultable long-term loans. In this model financial shocks affecting lending spreads can bring about a widespread recession that has at its core a deterioration in the equity of financial intermediaries and in their leverage capacity.

⁶For a recent paper suggesting a endogenous regime switching DSGE model to analyse financial crises in Mexico, see Benigno et al. (2020).

⁷Note that asset price driven cycles are more likely in market-based banking systems (IMF, 2009). The build-up of leverage is more likely in market-based systems due to the effective use of collateralization and sophisticated risk management and information-sharing strategies.

requirements as lessons from the GFC.

Financial constraints and economic dynamics: empirical evidence

Only a limited number of empirical contributions allow for nonlinearities in the relation between financial constraints and the macroeconomy, while a growing empirical literature has documented stylized facts on the role of financial factors in business cycles and for the development of a financial crisis. ⁸ Hubrich and Tetlow (2015) investigate whether a financial crisis is just a manifestation of amplified shocks or whether the transmission of shocks does actually change. They find empirical support for the hypothesis of a change in the transmission of financial shocks to the US macroeconomy in episodes of high stress. Hubrich et al. (2013) analyse the effects of financial shocks on the macroeconomy for EU and OECD countries and find evidence for nonlinearities and heterogeneity across countries in the transmission of financial shocks to the macroeconomy. Other studies also highlight empirical nonlinearities arguing that transmission channels may operate differently depending on underlying conditions, e.g. on the credit-to-GDP gap, for instance Aikman et al. (2020) or find different effects depending on the nature of the financial shocks, i.e. whether shocks represent easing or adverse financial conditions (see Barnichon et al. (2019)).

Brunnermeier et al. (2019) employing a structural VAR identified with heteroscedasticity, also find significant output effect of financial stress shocks, measured as spread shocks, including corporate bond (GZ) spread shock that is and an interbank lending spread shock as proxied by the 3-month Eurodollar rate over the 3-month Treasuries. A recent strand of literature studies the distribution of future real GDP growth as a function of current financial and economic conditions or bank capital using quantile regression. They find that the estimated lower quantiles of the distribution of future GDP growth exhibit strong variation as a function of current financial conditions (see for instance Adrian et al. (2019) and Boyarchenko et al. (2020)).⁹ We complement that literature by taking a parametric approach.

⁸Credit rises in the run-up to financial crises (Schularick and Taylor, 2012) and recessions associated with financial crises are usually deeper than normal recessions, especially if they are preceded with a build-up of credit (Jordà et al., 2013).

⁹See also Chavleishvili et al. (2021) for how to address macroprudential policy issues with a related approach.

3 The methodology

Most of the methodological literature focuses on models with constant probabilities of Markov switching. Following the seminal paper by Hamilton (1989), a number of contributions extended the basic Markov switching model and its estimation procedure suggested in that paper in different dimensions, see for instance Chauvet (1998), Kim and Nelson (1999), Frühwirth-Schnatter (2006), Sims and Zha (2006) and Sims et al. (2008).

Some papers propose a class of time-varying probability Markov switching regression models, including Filardo (1994), Diebold et al. (1994), Kim (2004), Kim et al. (2008) as well as Bazzi et al. (2017) and Chang et al. (2017). In these papers the probability of regime switching depends on certain variables of interest. They assume a functional form for the dependence of the probability on the state of the economy. Most papers employ a logistic function, sometimes a probit function is used (e.g. Kim et al. (2008)). Only a few recent papers allow for occasionally binding constraints in a VAR context that implies some endogeneity of regime switching, and those include Mavroeidis (2021), Aruoba et al. (2021) and Hayashi and Koeda (2019).¹⁰

In this paper, we propose a Regime-Switching Vectorautoregressive (RS-VAR) model with time-varying transition probabilities, building on and extending the framework presented in Sims et al. (2008).

We extend previous literature in several dimensions: 1. We allow for a time-varying transition matrix in a regime-switching structural VAR model; 2. we allow for a range of general, non-recursive identification schemes, including sign restrictions and narrative sign restrictions that might be different in different regimes; 3. We highlight and discuss identification issues in Regime-switching Structural VAR models.

3.1 The Regime-Switching Model with time-varying transition matrix

For $1 \le t \le T$, let y_t be an *n*-dimensional vector of endogenous variables, let z_t be a *k*dimensional vector of exogenous variables, and let s_t^c and s_t^v be a discrete latent variables with $s_t^c \in \{1, \dots, h_c\}$ and $s_t^v \in \{1, \dots, h_v\}$. We propose a structural vector autoregression with

¹⁰Two recent DSGE model proposals with endogenous switching can be found in Chang et al. (2018), and Benigno et al. (2020).

time-varying transition matrix (RS-SVAR)

$$A_0(s_t^c)y_t = A_+(s_t^c)x_t + \Xi^{-1}(s_t^v)\mathbf{\epsilon}_t,$$
(1)

where the predetermined vector x_t is $[y'_{t-1}, \dots, y'_{t-p}, z'_t]'$ and is of dimension m = np + k.¹¹ The exogenous structural shocks ε_t are *n*-dimensional and assumed to be standard normal and independent of the regime process s_t^c and s_t^v . The coefficient matrix $A_0(s_t^c)$ is $n \times n$ and invertible, $A_+(s_t^c)$ is $n \times m$, and $\Xi(s_t^v)$ is $n \times n$ and diagonal, with positive diagonal elements. We call s_t^c the coefficient regime and s_t^v the variance regime. We define the overall regime process to be $s_t = h_c(s_t^c - 1) + s_t^v$, which can take on $h = h_c h_v$ distinct values in $\{1, \dots, h\}$. The coefficient and variance regime processes are assumed to be independent, though this condition could be relaxed.¹²

We will denote the matrix of probabilities governing the transition of the processes s_t^c and s_t^v from time *t* to time t + 1 by $P_{t+1|t}^c$ and $P_{t+1|t}^v$, respectively. The matrix $P_{t+1|t}^c$ is $h_c \times h_c$ and the matrix $P_{t+1|t}^v$ is $h_v \times h_v$. The element in row *i* and column *j* of these matrices is the probability of transiting from regime *j* at time *t* to regime *i* at time t + 1. The elements of these matrices are all non-negative and the *columns* of each of these matrices must sum to one. In general, $P_{t+1|t}^c$ and $P_{t+1|t}^v$ can depend on the endogenous variables y_1, \dots, y_t , the exogenous variables z_1, \dots, z_t , and the matrices $A_0(\cdot), A_+(\cdot)$, and $\Xi(\cdot)$. This implies that $P_{t+1|t}^c$ and $P_{t+1|t}^v$ could also depend on the exogenous shocks $\varepsilon_1, \dots, \varepsilon_t$. In our empirical examples, the transition matrices will depend only on $y_{t-\ell}, \dots, y_t$, for some fixed non-negative value of ℓ .

In addition to the time-varying transition matrices, to fully specify the regime processes the initial probabilities must be specified. We denote these by p_0^c , which is an h_c -vector with non-negative elements that sum to one, and p_0^v , which is an h_v -vector with non-negative elements that sum to one.

The SVAR parameters of the model will be $A_0(\cdot)$, $A_+(\cdot)$, and $\Xi(\cdot)$. Both the transition matrices and the initial conditions can depend on the SVAR parameters and perhaps some additional vector of parameters that we will denote by q. The transition matrices, of course,

¹¹In our empirical application the only exogenous variable is a constant, $z_t = 1$.

¹²The assumption that s_t^c and s_t^v are independent is equivalent to the overall transition matrix from time *t* to time t + 1 being $P_{t+1|t} = P_{t+1|t}^c \otimes P_{t+1|t}^v$.

could also depend on the endogenous and exogenous data as described above. We will compactly represent all the parameters by $\theta = (A_0(\cdot), A_+(\cdot), \Xi(\cdot), q)$. For now, the only restrictions on the SVAR parameters are that the $A_0(\cdot)$ are invertible and the $\Xi(\cdot)$ are diagonal matrices with positive diagonal.

3.2 Identification in RS-SVAR models

One of the contributions of this paper is to highlight identification issues and to propose new identification schemes for RS-SVARs. Constant parameter SVAR models with homoskedastic Gaussian shocks are not identified, but constant parameter models with heteroskedastic shocks are identified, at least up to the ordering of the equations and sign of each equation, see Rigobon (2003). A similar result will hold for the RS-SVAR models we consider. For constant parameters structural VAR models with heteroskedastic shocks, the only identification issues are determining the ordering of the equations and the sign of each equation. For RS-SVAR models these identification issues are present, but there also two additional identification issues.

3.2.1 Identification through heteroskedasticity

Before stating our identification result, we need a restriction to pin down the relationship between $A_0(\cdot)$, $A_+(\cdot)$, and $\Xi(\cdot)$. If *D* is any diagonal matrix with positive diagonal, then the system given by Equation (1) and the system $DA_0(s_t^c)y_t = DA_+(s_t^c)x_t + D\Xi^{-1}(s_t^v)\varepsilon_t$ are observationally equivalent.¹³ Thus a restriction is needed to force *D* to be the identity. In the literature, some authors have chosen the restriction $\Xi(1) = I_n$. While this certainly solves the identification issue, it makes the first variance regime special. We will use the restriction $\sum_{k=1}^{h_v} \Xi^2(k) = I_n$, or $\sum_{k=1}^{h_v} \Xi^{-2}(k) = I_n$. This restriction treats all the variance regimes symmetrically and works better with the usual priors imposed on the $\Xi(\cdot)$. With this restriction, we have the following result.

Proposition 1 Suppose the span of the predetermined data is all of \mathbb{R}^m and the unconditional probability of being each overall regime is not zero for every t. If $h_v > 1$, then, for almost all

¹³It also must be the case that the new parameters, $DA_0(\cdot)$, $DA_+(\cdot)$, and $\Xi(\cdot)D^{-1}$, must satisfy the restrictions that $DA_0(\cdot)$ be invertible and $\Xi(\cdot)D^{-1}$ be a diagonal matrix with positive diagonal. Since *D* is diagonal with positive diagonal, both are satisfied.

parameters values, the RS-SVAR model given by Equation (1) is identified up to the ordering and sign of the equations and the ordering of the regimes.

Proof. See Appendix A.

The hypotheses of Proposition 1 are relatively mild. The first hypothesis is equivalent to the exogenous variables not being collinear and there being at least *m* observations The second hypothesis says that all the regimes are accessible. If all the initial probabilities are positive or if all the elements of the transition matrices are non-zero, then this hypothesis will be satisfied. Even if neither of these is true, as long at the positions of the zeros in the initial probabilities and the non-zero elements of the transition matrix do not exactly match up, then the hypothesis will be satisfied. In the next section, we discuss the precise meaning of the statement that the model is "identified up to the ordering and sign of the equations and the ordering of the regimes."

3.2.2 Other Identification Issues in RS-VAR models

As we saw in the previous section, multiplication of the system given by Equation (1) by an invertible matrix can result in an observationally equivalent system. In this section, we discuss the need for three more restrictions, all arising from multiplication of the system given by Equation (1) by an invertible matrix. This will make precise what we mean by "identified up to the ordering and sign of the equations and the ordering of the regimes." (see proposition 1).

If one were to permute the rows in Equation (1), then one would obtain an observationally equivalent system. More formally, if Q is a permutation matrix, then the system given by Equation (1) and the system $Q'A_0(s_t^c)y_t = Q'A_+(s_t^c)x_t + Q'\Xi^{-1}(s_t^v)QQ'\varepsilon_t$ are observationally equivalent.¹⁴ So, one must have a restriction that picks a unique ordering of the rows out of the n! possible ordering. Since the rows in Equation (1) correspond to equations and each equation contains a single shock, ordering the rows is referred to as ordering the equations or ordering the shocks. This is also referred to as identifying, or naming, the equations or shocks. For instance, we could order the equations so that the financial shock always appears in the first

¹⁴ If $\sigma(\cdot)$ is a permutation of $(1, \dots, n)$, then $Q = [e_{\sigma(1)}, \dots, e_{\sigma(n)}]$, where e_j is the *j*th column of the $n \times n$ identity matrix, is the *column* permutation matrix associated with $\sigma(\cdot)$. Permutation matrices are orthogonal and if *A* is any $n \times n$ matrix, then *AQ* permutes the columns *A* by $\sigma(\cdot)$ and *Q'A* permutes the rows of *A* by $\sigma(\cdot)$. So, if *Q* is a permutation matrix and *D* is a diagonal matrix, then *Q'DQ* permutes the diagonal elements of *D*. Thus, $Q'A_0(\cdot)$ is invertiable, $Q'\Xi^{-1}(\cdot)Q$ is a diagonal matrix with positive diagonal, and $Q'\varepsilon_t$ is standard normal.

equation. In this case we have identified or named the financial shock. If no such restriction is imposed, then we say the system given by Equation (1) is identified up to an ordering of the equations. The restrictions we will employ to order the equations will be discussed in Section 3.3.

If one were to multiply any equation in any coefficient regime in the system given by Equation (1) by minus one, then one would obtain an observationally equivalent system. More formally, if $D(k_c)$, for $1 \le k_c \le h_c$, is a diagonal matrix with plus or minus ones along the diagonal, then the system given by Equation (1) and the system given by $D(s_t^c)A_0(s_t^c)y_t =$ $D(s_t^c)A_+(s_t^c)x_t + \Xi^{-1}(s_t^v)\varepsilon_t$ are observationally equivalent. So, one must have a restriction to determine the sign of each equation in each coefficient regime. Such a restriction can be thought of as giving an interpretation to a positive shock. In the constant parameter case, Waggoner and Zha (2003) suggest a class of restrictions for determining the sign of each equation. For each coefficient regime, we will use a restriction from this class of restrictions. In particular, for each coefficient regime and equation, we will restrict the sign of that equation so that the impulse response of some variable at some horizon to a positive shock in that equation has a particular sign. In the above example, where the financial shock is ordered first, one could require that the contemporaneous response of output growth to a positive financial shock to be negative. In addition, one could require that the contemporaneous response of the financial conditions index to a positive financial shock to be positive. These would be consistent with a positive financial shock being detrimental to the outlook in growth. If no such conditions were imposed, then we say the system given by Equation (1) is identified up to sign.

In addition to identifying (or ordering) the equations, one must also identify (or order) both the coefficient and variance regimes. If $\sigma_c(\cdot)$ is a permutation of $(1, \dots, h_c)$ and $\sigma_v(\cdot)$ is a permutation of $(1, \dots, h_v)$, then we can define new discrete latent variables by $\tilde{s}_t^c = \sigma_c^{-1}(s_t^c)$ and $\tilde{s}_t^v = \sigma_v^{-1}(s_t^v)$. If Q_c is the $h_c \times h_c$ column permutation matrix associated with $\sigma_c(\cdot)$ and Q_v is the $h_v \times h_v$ column permutation matrix associated with $\sigma_v(\cdot)$, then the transition matrix and initial probabilities for \tilde{s}_t^c are $Q'_c P_{t+1|t}^c Q_c$ and $Q'_c p_0^c$ and the transition matrix and initial probabilities for \tilde{s}_t^v are $Q'_v P_{t+1|t}^v Q_v$ and $Q'_v p_o^v$. Furthermore, the system given by Equation (1) is observationally equivalent to the system given by $\tilde{A}_0(\tilde{s}_t^c)y_t = \tilde{A}_+(\tilde{s}_t^c)x_t + \tilde{\Xi}^{-1}(\tilde{s}_t^v)\varepsilon_t$, where $\tilde{A}_0(k_c) = A_0(\sigma_c(k_c)), \tilde{A}_+(k_c) = A_+(\sigma_c(k_c)), \text{ and } \tilde{\Xi}(k_v) = \Xi(\sigma_v(k_v)), \text{ for } 1 \le k_c \le h_c \text{ and } 1 \le k_v \le h_v.$ Thus we must have a restriction for selecting a unique ordering among the $h_c!$ possible orderings of the coefficient regimes and a restriction for selecting a unique ordering among the $h_v!$ possible orderings of the variance regimes. This can be thought of as identifying, or naming, each coefficient regime and each variance regime. If no such restrictions are imposed, then we say the system given by Equation (1) is identified up to an ordering of the regimes. The restrictions that we will employ for the ordering of the regimes will be discussed in Section 3.3.

3.3 Identifying Shocks and Regimes

Even in the constant parameter case with heteroskedastic shocks, coming up with plausible restrictions to identify the equations is difficult. This is one of the disadvantages of identification via heteroskedasticity. In the regime switching case, this difficulty is compounded because the coefficient and variance regimes must also be identified. In this section, we will discuss three different techniques for achieving both of these goals, sign restrictions, zero restrictions, and narrative restrictions. Combinations of these techniques can also be used. Last, we will discuss techniques for identifying the variance regimes.

3.3.1 Sign Restrictions

Sign restrictions on the impulse responses have long been used to identify the shocks in the case of constant parameters with homoskedastic shocks, though this identification is only set identification. In the case of regime switching parameters with heteroskedastic shocks, sign restrictions on the impulse responses can be used to identify both the equations and regimes. Even better, because there are only finitely many ways to identify the equations or regimes, in some cases sign restrictions can uniquely identify the equations or regimes, not just set identify. These ideas can best be explained by an example. For instance, suppose that one of the coefficient regimes is called the financial constraint regime and will be ordered first and that one of the shocks is the financial shock and will be ordered first. Note that the ordering of the equations must be the same across all coefficient regimes, so the financial shock would have to be ordered first in all the coefficient regimes. One could impose the restriction that the contemporaneous impulse response, conditional on being in the financial

constraint regime, to a positive financial shock is positive for both financial conditions index and leverage and negative for output growth and interest rates. For some parameter values, in no regime is there a shock whose impulse responses satisfy this pattern, so that parameter would be rejected. In this sense, multiple sign restrictions on the impulse responses to a given shock in a given regime imply that the model is overidentified.¹⁵ For other parameter values, there could be a unique regime and shock whose impulse response satisfied this pattern. In this case the sign restrictions uniquely determine the financial constraint regime and the financial shock and this regime and equation could be ordered first if that were not already the case. Finally, it could be the case that there are multiple regimes or shocks whose impulse response satisfied this pattern. In this case either the financial constraint regime or the financial shock, or both, would not be uniquely determined. As the number of sign restrictions increase, one would expect that the number of parameters that were rejected to increase and the number of parameters that did not uniquely determine both the regime and equation to decrease. If different sign restrictions on the impulse responses to the same shock across all the different regimes are imposed, then either the parameter will be rejected or the regimes will be uniquely determined. Similarly, if different sign restrictions on the impulse responses to all the different shocks, in any regime, are imposed, then either the parameter will be rejected or the equations will be uniquely determined. In this example, only one regime and one equation were being determined, though this idea could easily be extended to determining multiple regimes, i.e. partial identification, and equations or all of the regimes and equations

3.3.2 Zero Restrictions

Zero restrictions on the contemporaneous and predetermined parameters have also long been used to identify the shocks in the case of constant parameters with homoskedastic shocks. These ideas can also be used to identify the regimes and equations. Given a coefficient regime, if the pattern of zero restrictions on the contemporaneous and predetermined parameters in that regime is different from the pattern in all other coefficient regimes, then the given coefficient regime is uniquely determined by the zero restrictions, for almost all parameter values. Given

¹⁵A single sign restriction on the impulse responses to a given shock in a given coefficient regime does not impose overidentifying restrictions because the sign of any equation in any coefficient regime could be changed. Alternatively, as we saw in Section 3.2.2, a single sign restriction on the impulse response to a given shock in a given regime could be thought of as a restriction determining the sign of the given equation in the given regime.

an equation, if for all other equation there exists a coefficient regime such that the patterns of zero restrictions in those two equations differ in that regime, then the given equation is uniquely determined by the zero restrictions, for almost all parameter values. In the case of heteroskedastic shocks, a single zero restriction overidentifies the model. Using different patterns of zero restrictions across different equations and coefficient regimes is a powerful way of identifying the equations and regimes. However, such restrictions can severely overidentify the model resulting in a poor fit to the data. They should be used judiciously and in combination with sign and narrative restrictions.

3.3.3 Narrative Restrictions

Narrative restrictions were used by Antolín-Díaz and Rubio-Ramírez (2018) to set identify shocks in the case of constant parameters with homoskedastic shocks. Even earlier, Sims and Zha (2006) used similar ideas to identify the regimes in a Markov switching monetary policy model. We describe this procedure in the context of our model. In the middle of September in 2008, Lehman Brothers failed. Our contention is that by October of 2008, there was a high probability that the economy was in what we will call the financial constraint regime, which we will order first. We can use this to uniquely determine the financial constraint regime. In our case, since there are only two coefficient regimes, if we can determine the financial constraint regime, then the other regime, which we call the normal regime, will also be uniquely determined. The first step is to define what we mean when we say that there is high probability that we are in coefficient regime k at time t. To do this one must choose a cutoff probability, and if the smoothed probability that we are in coefficient regime k at time t is greater than the cutoff probability, then we say we there is a high probability that we are in coefficient regime k at time t. If the cutoff probability is greater than or equal to 0.5, then there is either a unique coefficient regime that is of high probability at time t or no coefficient regime that is of high probability at time t. Among the parameter values for which there is a unique coefficient regime that is of high probability in October of 2008, we will order the coefficient regimes so that this regime is first and call the first regime the financial constraint regime. If one was unsure of the exact period that economy was in the financial constraint regime, then one could choose a window about October of 2008, and then say the financial

constraint regime was the regime that was in high probability over most of this window. For our models, we found that neither the choice of window about October of 2008 nor the cutoff probability, within reason, affected the determination of the financial constraint regime.

Similar ideas could be used to identify the equations. In our model, we wish to uniquely determine what we will call the financial shock. Again, in October of 2008, there was a massive decline in industrial production and it is our contention that the financial shock caused most of this decline. We first must define what we mean by when we say that "shock k caused most of the decline in industrial production at time t". For each overall regime, one can compute the expected value of each time t shock, conditional on the overall regime at time t. One could then compute the expected contemporaneous impulse response of industrial production to each time t shock, conditional on the overall regime at time t. Finally, using the smoothed probabilities, one could then compute the expected contemporaneous impulse response of industrial production to each time t shock. The time t shock with largest negative expected contemporaneous impulse response of industrial production caused most of the decline in industrial production at time t. Except in knife edge cases, there is a unique shock that caused most of the decline in industrial production in October of 2008 and we will order the equations so that shock is first and call the first shock the financial shock. As with ordering the regimes, if one was unsure of the exact period that the financial shock was dominant, then one could choose a window about October of 2008, and then say the financial shock was the shock that caused the most cumulative decline in industrial production over this window.

3.3.4 Identifying the Variance Regimes

In some ways, identifying the variance regimes is more straightforward. The inverse of the diagonal elements of $\Xi(\cdot)$ directly scale the structural shocks, and thus have an economic interpretation. For instance, in our models we are interested in the financial shock, which we order first. So, we order the variance regimes, after we have ordered the coefficient regimes, so that the first diagonal element of the $\Xi(\cdot)$ are in increasing order. This would imply that the first variance regime would have the most variability, at least in terms of the effect of the financial shock. We should point out that under this ordering, in the first variance regime the impulse response to a financial shock will have the largest response for all variables at all

horizons in all coefficient regimes.

3.4 The Transition Matrices and Initial Probabilities

For the variance regime we will use a constant transition matrix and for the coefficient regime we will have time-varying transition matrices of a particular functional form. Our methodology will certainly allow for both the coefficient and variance regimes to have time-varying transition matrices of completely general functional forms, but in the interest of parsimony, we restrict the variance regime to have a constant transition matrix and the diagonal elements of the coefficient regime transition matrix to be a logistic transformation of a linear function of the endogenous variables. The off-diagonal elements of the coefficient regime transition matrix will be a constant times one minus the diagonal element from the same column. For $1 \le i, j \le h_v$, we will denote the constant elements in the variance regime transition matrix by $q_{i,j}^v$ and note that $\sum_{i=1}^{h_v} q_{i,j}^v = 1$. Let q_v denote the vector containing all the $q_{i,j}^v$.

Let $p_{t+1|t}(i, j)$ denote the time-varying probability of switching from regime *j* at time *t* to regime *i* at time t + 1. We assume that the time-varying probability of staying in the *j*th regime at time t + 1, given that we are in the *j*th regime at time *t*, is of the form

$$p_{t+1|t}(j,j) = \frac{1}{1 + \exp(\bar{\gamma}_j - \sum_{k=1}^{\ell} \gamma_{j,k} y_{t-k+1})}.$$
(2)

The scalars $\bar{\gamma}_j$ are the location parameters and the *n*-vectors $\gamma_{j,k}$ are the slope parameters. In keeping with our desire to be parsimonious, in most of our examples, $\ell = 1$ and only a few of the elements of $\gamma_{j,k}$ will be allowed to be non-zero. We will gather all of these parameters that are not restricted to zero into a vector that we will denote by γ .

In the case of only two coefficient regimes, the diagonal elements completely determine the transition matrix. If there are more than two coefficient regimes, then for $i \neq j$,

$$p_{t+1|t}(i,j) = q_{i,j}^c (1 - p_{t+1,t}(j,j)),$$
(3)

where the $q_{i,j}^c$ are non-negative constants such that $\sum_{i=1}^{h_c} q_{i,j}^c = 1$, under the convention that $q_{j,j}^c = 0$. Let q_c denote the vector containing all the $q_{i,j}^c$ that are not restricted to be zero.

We will choose the initial probabilities in both the coefficient and variance regime pro-

cesses so that all the regimes have equal probability. This choice is mandated by the fact that we want the initial probabilities to be invariant to permutations of either of the coefficient or variance regime. If this was not the case, then the initial probabilities would determine, at least partially, the ordering of the coefficient and variance regimes. Unless one was very sure about the initial regime, this would not likely result in a satisfactory condition for ordering the coefficient and variance regimes.

In the case of the variance regime, which has constant transition matrix, choosing the initial variance probabilities to be the ergodic probabilities would also be invariant to permutations of the variance regime. This would be a permissible choice and not increase the number of parameters. However, in most cases this would not deliver substantially different results.

There are no parameters controlling the initial probabilities and the parameters controlling the transition matrices are (q_v, q_c, γ) . As stated before, we gather all of these parameters into a vector that we will denote by q.

3.5 The Priors

In this section we describe the priors that we will employ.

For each $1 \le k_c \le h_c$, we will use the same Sims-Zha prior on each $(A_0(k_c), A_+(k_c))$. For the hyperparameters in this priors, we will follow the recommendations in Sims and Zha (1998) for monthly data.

For each $1 \le k_v \le h_v$, denote the diagonal elements of $\Xi(k_v)$ by $\xi_{k_v,j}$, for $1 \le j \le n$. Recall that we use the restriction that either $\sum_{k_v=1}^{h_v} \xi_{k_v,j}^2 = 1$ or $\sum_{k_v=1}^{h_v} \xi_{k_v,j}^{-2} = 1$, for each $1 \le j \le n$. We will use the uniform prior across the $\xi_{k_v,j}$. This can be easily implemented as a Dirichlet distribution over $(\xi_{1,j}, \dots, \xi_{h_v,j})$, for each $1 \le j \le n$, with all the Dirichlet hyperparameters equal to one.

For the variance regime transition matrices, the parameters are $q_{i,j}$, for $1 \le i, j \le h_v$. We will use a Dirichlet prior on $(q_{1,j}^v, \dots, q_{h_v,j}^v)$, for each $1 \le j \le h_v$. For the off-diagonal elements, we will choose the Dirichlet hyperparameters to all be equal to one. For the diagonal elements, we will choose the hyperparameter to match the desired duration of each variance regime, though we will assume that the duration is the same across all variance regimes.

For the coefficient regime transition matrices, the parameters are (q_c, γ) . We will use a

Dirichlet prior on $(q_{1,j}^c, \dots, q_{j-1,j}^c, q_{j+1,j}^c, \dots, q_{h_{\nu},j}^c)$, for each $1 \le j \le h_{\nu}$. We will choose the Dirichlet hyperparameters to all be equal to one, so the distribution will be uniform. For the γ , we will use independent normal distributions. We recommend standardizing all the variables controlling the diagonal elements of the coefficient regime transition matrices. In this case we find that using independent normal distributions works well. Alternatively, if one had prior opinions about the means and variances of the variables controlling the diagonal elements of the variables about the means and variances of the variables controlling the diagonal elements of variances of the variables controlling the diagonal elements of variances of the variables controlling the diagonal elements of variances of the variables controlling the diagonal elements of variances of the variables controlling the diagonal elements of variances of the variables controlling the diagonal elements of variances of the variables controlling the diagonal elements of variances of the variables controlling the diagonal elements and variances of the variables controlling the diagonal elements of variances of the variables controlling the diagonal elements and variances of the variables controlling the diagonal elements and variances of the variables controlling the diagonal elements of the coefficient regime transition matrices, then these could be used to set the means and variances of the elements of γ .

Because we assume that the initial probabilities are all equal, there are no parameters associated with the initial parameters.

3.6 The Posterior, Filtered and Smoothed Probabilities

In this section, we give formulas for the posterior, filtered and smoothed probabilities. For completeness, these formulas will be explicitly derived in Appendix B. To derive expressions for the posterior and filtered probabilities, we need the following assumption about the exogenous variables.

Assumption 1 (Exogeniety) $p(z_{t+1}|S_t, Y_t, Z_t, \theta) = p(z_{t+1}|Y_t, Z_t)$, where $S_t = (s_1, \dots, s_t)$, $Y_t = (y_1, \dots, y_t)$, and $Z_t = (z_1, \dots, z_t)$.

The idea is that $p(z_{t+1}|Y_t, Z_t)$ is the true, but unknown, distribution of z_{t+1} , conditional on Y_t and Z_t , and knowing the path of regimes or the model parameters provides no additional information. Note that this assumption implies that the conditional distribution of z_t does not depend on the regimes. Under this assumption, if $p(\theta)$ is the prior, then the posterior is proportional to

$$P(\theta|Y_T, Z_T) \propto p(\theta) \prod_{t=1}^T \sum_{s_t=1}^h p(y_t|s_t, Y_{t-1}, Z_t, \theta) p(s_t|Y_{t-1}, Z_{t-1}, \theta).$$
(4)

In order to compute the posterior, we must be able to explicitly compute the conditional likelihood, $p(y_t|s_t, Y_{t-1}, Z_t, \theta)$, and the filtered probabilities, $p(s_t|Y_{t-1}, Z_{t-1}, \theta)$. For RS-SVAR models, the conditional likelihood is normal and easy to compute. Given the initial probabilities, the filtered probabilities can be recursively computed via the Hamilton filter. The recursive

formulas are

$$p(s_{t+1}|Y_t, Z_t, \theta) = \sum_{s_t=1}^h P_{t+1|t}(s_{t+1}, s_t) p(s_t|Y_t, Z_t, \theta)$$
(5)

$$p(s_t|Y_t, Z_t, \theta) = \frac{p(y_t|s_t, Y_{t-1}, Z_t, \theta)p(s_t|Y_{t-1}, Z_{t-1}, \theta)}{\sum_{s_t=1}^h p(y_t|s_t, Y_{t-1}, Z_t, \theta)p(s_t|Y_{t-1}, Z_{t-1}, \theta)}$$
(6)

Often, one is more interested in the smoothed probabilities, $p(s_t|Y_T, Z_T, \theta)$. This can be done using backward recursion in the Hamilton smoother. The formula for the backward recursion is

$$p(s_t|Y_T, Z_T, \theta) = p(s_t|Y_t, Z_t, \theta) \sum_{s_{t+1}=1}^h \frac{p(s_{t+1}|s_t, Y_t, Z_t, \theta) p(s_{t+1}|Y_T, Z_T, \theta)}{p(s_{t+1}|Y_t, Z_t, \theta)}.$$
(7)

Since $p(s_T|Y_T, Z_T, \theta)$ can be obtained from the last step of the Hamilton filter, we can start the backward recursion at s_T and then recursively compute the smoothed probabilities.

3.7 Estimation

Having efficient and accurate samplers for simulating the posterior distribution is crucial for Bayesian analysis. There are a number of recent papers proposing new methods to compute posterior distributions, e.g. Durham et al. (2014), Herbst and Schorfheide (2014), Bognanni and Herbst (2018), and Waggoner et al. (2016).

We employ the dynamic striated Metropolis-Hastings (DSMH) sampler (see Waggoner et al. (2016)) [henceforth WWZ (2016)] to our new model class to derive the whole posterior distribution. The DSMH sampler is grounded in the Metropolis-Hastings algorithm, it pools the strengths from the equi-energy and sequential Monte Carlo samplers while avoiding the weaknesses of the standard Metropolis-Hastings algorithm and those of importance sampling. The basic idea of the Dynamic Striated Metropolis-Hastings sampler is to move from a tractable initial distribution one can sample from and transform the initial distribution gradually to the desired posterior distribution through a sequence of stages. Dynamic adjustment is used for sampling form the target distribution at each stage when progressing to the next stage. The sampler utilizes importance weights only for initial draw at each stage to avoid the degeneracy problem of other Sequential Monte Carlo samplers.¹⁶

4 Empirical Analysis

4.1 Motivation

We complement previous empirical studies on financial constraints and economic dynamics by providing empirical evidence on the role of leverage of financial institutions for the transmission of financial shocks to the macroeconomy using our proposed regimes switching model with new identification techniques. Our motivation to focus on leverage is threefold: First, the recent theoretical literature emphasizes the role of bank balance sheets for the build-up of financial instabilities and the amplification of economic downturns. Second, leverage encompasses the entire balance sheet of the financial institution and therefore is a broad indicator for signaling financial vulnerabilities. Third, we aim to contribute to the discussion regarding the usefulness of the leverage ratio from a financial stability policy perspective.¹⁷

We empirically investigate the role of balance sheets of financial institutions for the amplification of financial shocks, differences in the transmission of financial shocks in different regimes, and the heterogeneity of financial institutions and implications for the persistence of financial constraint regimes. We build on Adrian and Brunnermeier (2016) and Paul (2019) and construct a novel market measure of leverage of financial institutions. We employ our proposed regime switching structural vector autoregressive (SVAR) model for this investigation.

4.2 Data, model specification and identification

4.2.1 Data

Leverage of financial institutions is an important regulatory indicator of financial sector vulnerabilities, since highly levered financial institutions are less able to absorb losses. Recent literature on structural macroeconomic models emphasizes the role of bank balance sheets for

¹⁶We use the following settings: 50 stages, 50 striations, 100 Groups, 2000 draws within each group; overall, we have therefore 200000 draws.

¹⁷Note that the supplementary leverage ratio has been introduced in the US in 2014. The aim is to counterbalance the build-up of systemic risk by limiting the risk weights compression (downweighting of seemingly low risk investments) during booms and therefore add a more countercyclical measure than a risk weighted capital ratio. For a discussion, see e.g. Gambacorta and Karmakar (2018).

the build-up of financial instabilities and the amplification of economic downturns. Furthermore, leverage encompasses the entire balance sheet of the financial institution and therefore is a broad indicator for signaling financial vulnerabilities.¹⁸

In our empirical analysis we use a market value, micro-data based measure of leverage of financial institutions, building on Adrian and Brunnermeier (2016). We employ the CRSP/Compustat merged database that covers a broad range of publicly listed depository and nondepository institutions, bank holding companies and non-banks.

Market leverage is constructed using market value equity, not the book value, i.e. based on the expected present discounted value of future cash flows of a financial institution, its creditors and its shareholders; in contrast, book values depend on specific accounting rules. Therefore, the leverage measure with market equity that we are using here takes into account that before the global financial crisis the leverage of financial institutions only rose mildly, since as debt went up also the market value of asset prices increased. During the crisis asset prices collapsed, while financial institutions were not able to reduce their accumulated debt burden as quickly leading to a sharp increase in leverage using the market value of equity. Note that our database includes all listed financial institutions, including a broad range of depository and non-depository credit institutions and a range of non-bank institutions, including security brokers and dealers. We use data from the Fundamentals Quarterly and Security Monthly of the CRSP/Compustat Merged database. We compute a novel monthly market leverage measure based on monthly market equity and quarterly interpolated series for book assets, or - alternatively - liabilities. Our measure builds on Adrian and Brunnermeier (2016), but goes beyond previous literature that uses linear interpolation to convert quarterly book values. We employ monthly call reports data for interpolation of the quarterly book assets and book liabilities. The source of the data used for monthly interpolation of the quarterly book assets and liabilities is a monthly survey of a sample of the commercial banks in the call reports.¹⁹ We compute leverage as book assets over market equity as well as market equity plus book

¹⁸Under Basel III, the supplementary leverage ratio is introduced as a measure that treats all exposures equally, independent of any risk assessment. The non-risk weighted leverage ratio is intended to avoid that banks lever up their balance sheet by investing in assets that appear in low-risk categories.

¹⁹The H.8 data from the Federal Reserve presents an estimate of weekly aggregate balance sheets (assets and liabilities) of commercial banks in the United States. The data are based on weekly reports from 875 commercial banks.

liabilities over market equity. ²⁰

Recent literature has highlighted that book and market leverage diverge substantially during crises (see also Begenau et al. (2021). We provide evidence that market leverage is a useful indicator for monitoring financial institutions since it reflects market developments in a timely way. The advantage of using financial institutions level data also is that they allow us to analyze the economic implications of heterogeneity across financial institutions.²¹ In particular, we focus on comparing model specifications with leverage of depository institutions, with leverage of Globally Systemically Important Banks (GSIBs) and with leverage of a particular group of non-bank financial institutions, namely securities brokers and dealers.

In addition to this novel monthly measure of market based leverage, we use US monthly data, seasonally adjusted, for a sample from 1988(12) to 2019(12). We include the following variables: Output growth is measured in terms of growth in industrial production, we include core CPI inflation and the 2-year Treasury rate; market leverage of financial institutions or leverage of particular financial institutions such as Globally Systemically Important Banks (GSIBs) and security brokers and dealers. Market leverage is proxied by book assets over market equity, as described above, and we include a broad financial conditions index including spreads, financial market volatility measures and other financial conditions indicators spanning a broad range of financial markets and financial intermediaries, since we are interested in investigating heterogeneity. It is published by the Federal Reserve Bank of Chicago. We chose this broad measure of financial conditions since we are interested in investigating and comparing the role of heterogeneity of leverage of financial institutions for economic outcomes.

4.2.2 Model specification

We employ our proposed regime switching vector autoregressive (RS-VAR) model with timevarying transition probabilities. We allow for two regimes in the VAR coefficients, which

²⁰The latter is often referred to as being a more reliable measure of market leverage since book liabilities are a better proxi for market liabilities than book assets for market assets, and that is what we use for our main empirical analyses.

²¹Note that the treatment of mergers and acquisitions in Compustat is as follows: When firms merge, the financial balance sheet items of the target firm gets absorbed into the balance sheet items of the acquirer. Therefore, when the target firm's data series ends, the acquirer's data series reflects the target financial balance sheet items. This provides the background for how structural changes in the course of the GFC will be reflected in this data set.

we label 'financial constraint' regime and 'normal' times. We make the regime probability dependent on the financial variables in the model since we are primarily interested in the transmission of financial shocks. We also allow for two regimes for the variances that follow a Markov process as a way to model heteroskedasticity.

In our regime switching models with time-varying probabilities - that can also be referred to as "endogenous switching" models - the transition probability of being in one regime in the next period, given that we are in a particular regime in this period, can vary over time. We model the transition matrix to depend on the state of the economy, namely we make it dependent on the financial variables in our system. To identify the regimes and structural shocks, we employ sign restrictions and, alternatively, narrative sign restrictions, thereby extending the approaches suggested in Antolín-Díaz and Rubio-Ramírez (2018) and Arias et al. (2018) to regime switching models. We also allow for different identification schemes in different regimes. We explain the details in the next section.

4.2.3 Identifying Regimes and Shocks

Regime identification In Section 3.3, we discussed how to identify the regimes using narrative restrictions. In this section we give the specific details of how we assigned the regimes to either the financial constraint regime or the normal regime. For each regime, we counted the number of months between 2008(9) and 2009(8), inclusive, that the probability of being in that regime was greater than 0.70. Whichever regime had the larger count was labeled the 'financial constraint' regime and ordered first. The other regime was labeled the 'normal' regime and ordered second. The assignment of the financial constraint regime was robust to varying the cutoff probability from 0.50 through 0.95 and choosing a shorter period around the Global financial crisis, or even only using 2008(10). Put in another way, the data was very informative and clear as to which regime the economy was in during the Global financial crisis.

Shock identification In addition to identifying the regimes, in our RS-VAR model we also need to identify the shocks. In particular, we want to identify the financial shock, which we will order first. We used sign restrictions on the impulse responses to achieve this. The contemporaneous response to a positive financial shock in the financial constraint regime was

restricted to be negative for output, inflation and short-term interest rate, but positive for the financial conditions index and leverage. The contemporaneous response to a positive financial shock in the normal regime was restricted to be positive for the financial conditions index only. Overall, in about 20 percent of the draws these sign restrictions uniquely identified the financial shock.

Alternative shock identification We have carried out further empirical analyses using narrative restrictions as an alternative shock identification approach. This is a new class of narrative restrictions developed for constant parameter structural VAR models (see Antonlin-Diaz and Rubio-Ramirez, 2018) that we extend to the regime switching structural VAR model setup. The structural parameters are constrained in such a way around key historical events that structural shocks and historical decomposition agree with the narrative. Those narrative sign restrictions combine the appeal of narratives with the advantages of sign restrictions. A small number of key historical events (not whole time series) are used for identification, thereby avoiding measurement error in narrative time series that have been presented in earlier literature.

In our implementation we identify the financial shocks as the one that explains most of the variation in output growth during the Global financial crisis. In addition, a considerably smaller set of sign restrictions are imposed than were when only sign restrictions were used. We only impose the positive response of the financial conditions contemporaneously, as well as the negative initial output growth response and the initial positive response of leverage. We present the empirical results using standard sign restrictions as our baseline results in the next few sections, and show some key results for narrative sign restrictions in Section 4.8 and Appendix C.

4.3 **Regime probabilities**

In the following we use smoothed probabilities as well as the time-varying probabilities and their association to historical events to interpret the different regimes. We use the specification with market leverage of GSIBs as our baseline specification and discuss those results first. We choose a model specification with endogenous regime switching driven by the three financial variables in our model, namely the 2-year Treasury rate, leverage and financial conditions.²² This choice is motivated by our interest in the financial shock transmission.

Figure 1 presents the smoothed probability of the first coefficient regime allowing for both variance regimes based on our endogenous regime switching specification. We present the estimate of the median of the posterior distribution.²³ We interpret this regime as a "financial constraint" regime; it covers the end of the Savings & Loan crisis, the 1990/91 recession, the Russian debt default, the GFC and the related recession. Note that the filtered probabilities - that are particularly useful for determining the financial constraint regime in real-time - are presented in Figure 2 and are very similar to the smoothed probabilities. We will use the impulse responses to a financial shock presented in the next section to shed light on the economic dynamics in this regime.





Figure 2: Filtered probabilities of coefficient regime



²²This model allows for four regimes given the different combinations of variance and coefficient regimes. We focus on one coefficient regime allowing for both variance regimes to occur during this regime in what is presented in the charts.

²³Note that the error bands for the smoothed probabilities are rather tight around the median.

Figure 3: Time-varying probability of staying in low interest rate regime, conditional of being in that regime



Figure 3 displays the time-varying probability of staying in the financial constraint regime, conditional on being in that regime (the latter probability as indicated in Figure 1). The figure illustrates that the probability of staying in the financial constraint regime is one at the end of 2008 and at the beginning of 2009, but declines sharply in 2009/2010 down to nearly 0.6 percent, increasing again during the European sovereign debt crisis before declining again further.²⁴ This figure illustrates the value of a model with endogenous regime switching for economic interpretation. Time-varying probabilities with error bands for the specification with GSIBs' leverage are presented in Appendix D and illustrate that uncertainty around this probability of staying in the financial constraint regime is small during the GFC and even thought it increases when the probability is declining the decline, subsequent rise and further decline are significant.

4.4 The transmission of financial shocks

An important contribution of our paper is to shed light on the role of leverage of financial institutions for the transmission of financial shocks to the real economy. Our model specification allows for two coefficient regimes that depend endogenously on the financial variables in the system and for two variance regimes. We now discuss the transmission of a financial shock starting with the financial constraint regime and then compare the results to normal times shock responses.

²⁴Note that the time-varying probability of staying in a regime can only be interpreted during the time when there is a high probability of being in that regime corresponding to the smoothed probability depicted in the previous Figure.

First, we investigate the impulse responses to a financial shock in our model when including leverage of the GSIB institutions under supervision of the Federal Reserve Board.²⁵ Figure 4 presents the impulse responses to a one standard deviation financial shock in the financial constraint regime (for one particular variance regime) conditional on staying in the financial constraint regime. In other words, we assume that the economy stays in the financial constraint regime for 12 months after the shock has hit. The median response of the whole posterior distribution is displayed with 68% error bands.

Figure 4: Impulse Responses to a financial shock in the financial constraint regime, GSIBs



Note: *IRFs* conditional on regime; red line: median response, blue lines: lower and upper bound of the 68 percent error bounds; financial shock: 1 std shock to financial conditions index; identification with contemporaneous sign restrictions, median and 68% error bands, output growth (IP), core inflation (CPI), 2-y Treasury yield, market leverage, Financial conditions index (Chicago Fed)

The impulse responses show an output response that turns out to be significantly negative, large and protracted, as would have been expected in the financial constraint regime. We find that leverage of GSIBs initially increases due to a sharp decline in asset prices, and then starts declining since the GSIBs deleverage in response to a financial shock in financial constraints episodes. We interpret that as evidence that deleveraging can lead to amplification effects with adverse implications for the real economy. GSIBs deleverage by liquidating assets, for

²⁵Capitalization of GSIBs are regularly monitored in the Financial Stability Report published by the Federal Reserve. They include Bank of America Corporation, the Bank of New York Mellon Corporation, Citigroup Inc., The Goldman Sachs Group, Inc., JP Morgan Chase Co., Morgan Stanley, State Street Corporation, Wells Fargo Company.

instance by carrying out (fire) sales of securities and/or extending fewer loans while existing loans mature. This implies a reduction in overall credit supply. At the same time, it will be more demanding to get external financing due to a decline in collateral value.

Next, we compare the responses of the economy in the financial constraint regime and in normal times. Figure 5 shows the responses to a financial shock in normal times. We find that output growth shows a large negative response in normal times, but that response is non-persistent in contrast to the financial constraint regime. Also, in contrast to the financial constraint regime market leverage remains insignificant over the entire horizon.

Figure 5: Impulse Responses to a financial shock in normal regime. GSIBs



Note: *IRFs* conditional on regime; red line: median response, blue lines: lower and upper bound of the 68 percent error bounds; financial shock: 1 std shock to financial conditions index; identification with contemporaneous sign restrictions, median and 68% error bands, output growth (IP), core inflation (CPI), 2-y Treasury yield, market leverage, Financial conditions index (Chicago Fed)

Since these impulse responses provide an average response for the respective regime and are conditional on the regime, we shed more light on the role of leverage during the GFC using a counterfactual analysis in Section 4.6. Before that we turn to investigating the role of the heterogeneity of financial institutions in terms of leverage for the transmission of financial shocks.

Heterogeneity of financial institutions: Leverage of Depository Institutions

Our second set of results is for depository financial institutions (including commercial banks, savings and loans, and credit unions)²⁶ from our CRSP/COMPUSTAT database of listed institutions. Again, sign restrictions are only imposed contemporaneously on all endogenous variables in response to a positive financial shock in one of the regimes, and only on one shock in the other regime.

In comparison to the results for the GSIBs our findings for Depository Institutions display a similar median output growth response for a given financial conditions tightening (Figure 6). However, the results do not display asymmetric responses in the tails of the posterior distribution as for the GSIBs, and hence - in contrast to the model with GSIBs leverage negative growth outcomes appear not to be more likely than positive outcomes. We also find that, as for GSIBs (see Figure 4), market leverage initially increases significantly due to the sharp decline in asset prices and then starts declining gradually since financial institutions deleverage.

The responses to a financial shock in normal times (Figure 7) show large and non-persistent output growth effects and no significant leverage response, similar to the responses we saw for GSIBs (Figure 5).

4.5 Heterogeneity: Securities brokers and dealers

Next we examine the role of heterogeneity by including leverage of security brokers and dealers in the model.

The financial constraint regime is again identified by sign restrictions only contemporaneously on all endogenous variables in one of the regimes, as for previous specifications. As pointed out in Aramonte et al. (2021) higher leverage does not need to correspond to larger balance sheets, but broker dealers clearly used debt to finance asset growth (see Adrian and Shin (2014)).

We find similar results for the real economy implications as before, with a protracted negative output growth in response to a financial shock, with asymmetric responses in the tails

²⁶Depository institutions are financial institutions that receive money from depositors to lend out to borrowers such as commercial banks and the other institutions listed here.

Figure 6: Impulse Responses to a financial shock in the financial constraint regime, depository institutions



Note: *IRFs* conditional on regime; red line: median response, blue lines: lower and upper bound of the 68 percent error bounds; financial shock: 1 std shock to financial conditions index; identification with contemporaneous sign restrictions, median and 68% error bands, output growth (IP), core inflation (CPI), 2-y Treasury yield, market leverage, Financial conditions index (Chicago Fed)

Figure 7: Impulse Responses to a financial shock in normal regime, depository financial institutions



Note: *IRFs* conditional on regime; red line: median response, blue lines: lower and upper bound of the 68 percent error bounds; financial shock: 1 std shock to financial conditions index; identification with contemporaneous sign restrictions, median and 68% error bands, output growth (IP), core inflation (CPI), 2-y Treasury yield, market leverage, Financial conditions index (Chicago Fed)

Figure 8: Impulse Responses to a financial shock in the financial constraint regime, securities brokers and dealers



Note: *IRFs* conditional on regime; red line: median response, blue lines: lower and upper bound of the 68 percent error bounds; financial shock: 1 std shock to financial conditions index; identification with contemporaneous sign restrictions, median and 68% error bands, output growth (IP), core inflation (CPI), market leverage = assets/market equity, Financial conditions index (Chicago Fed)

of the posterior distribution as for the GSIBs leverage specification. However, market leverage increases more on impact than for other financial institutions and then immediately declines due to deleveraging. We interpret that as reflecting that The dealers' willingness to take risk amplified the growth of the dealer balance sheets going into the crisis, causing crisis losses and a subsequent sharp contraction of balance sheets post-crisis.

4.6 Counterfactual: Leverage of GISBs

To shed further light on the role of GSIBs leverage in the financial constraint regime, we carry out a counterfactual zooming into the GFC. We simulate the data based on what would have happened if (market) leverage of GSIBs would have remained constant as of October 2008. This experiment is designed to focus on the deleveraging process and its potential adverse effects on the real economy, by avoiding the final sharp increase of market leverage at the onset of the GFC and subsequent fall. As illustrated in Figure 9 this would have implied a less pronounced tightening of the financial conditions in the Fall of 2008 and subsequent quicker recovery. Also, this counterfactual would have implied a dramatically smaller decline





Note: Probability of staying in financial constraint regime, red line: counterfactual data with constant GSIBs leverage, black line: actual data; Output growth (IP), core inflation (core CPI), 2-y Treasury yield, leverage (market-based, GSIBs), financial conditions index (Chicago Fed)

in output growth (about 20 percentage points in terms of growth in industrial production) and a subsequent quicker recovery in output growth and related inflation increases despite higher short-term interest rates.

Counterfactual probability of staying in the financial constraint regime: Leverage of GISBs versus Depository Institutions

We also compute the counterfactual probability of staying in the financial constraint regime for the case of keeping leverage of GSIBs constant.

Figure 10 shows that the counterfactual probability of staying in the financial constraint regime (right panel) declines more than the actual probability (left panel). Some of the outcomes indicate a much lower probability of staying in the financial constraint regime under the counterfactual scenario of constant leverage, namely as low as 0.1. This is in line with what might be expected since the median output response might be similar than for other financial institutions, but outcomes might be more detrimental in response to financial shocks for GSIBs that are particularly highly leveraged.

Indeed, this is what is illustrated in Figure 11 for a system including depository financial

Figure 10: Counterfactual probability of staying in the financial constraint regime, GSIBs



Note: Black line: median time-varying probability of staying in financial constraint regime, black line: upper error bound, red line: lower bound; left panel: actual, right pane: counterfactual

institutions' leverage where the lower bound probability of staying in the financial constraint regime does not go down below 0.90.

4.7 Market leverage and financial conditions

To illustrate that market leverage and financial conditions take distinct roles for the transmission of financial shocks, we have carried out a number of counterfactuals. We hold the financial conditions index constant as of October 2008 (to make it similar in terms of timing to our other experiment that focuses on the deleveraging process) and compute the counterfactual probability of staying in the financial constraint regime in the model with leverage of GSIBs.

The counterfactual probability of staying in the constraint regime is not going down as in the case of holding leverage constant. We interpret this as evidence that NFCI and leverage provide different characterizations of the financial conditions of the economy and have different implications for the propagation of shocks and the persistence of the constraint regime. It also illustrates that it is not the market price that is the sole driver for our results, since that would be behind both the financial conditions index and leverage.

We have also estimated our model that includes leverage of GSIBs with the GZ spread instead of the the broad financial conditions index to illustrate that it is not the leverage measures

Figure 11: Counterfactual probability of staying in the financial constraint regime, depository financial institutions



Note: Black line: median time-varying probability of staying in financial constraint regime, black line: upper error bound, red line: lower bound; left panel: actual, right pane: counterfactual

or stock market variables related to financial institutions that are behind our results. We get very similar results in that the probability of staying in the constraint regime is declining much more than the actual probability, confirming our result that it helps to prevent the deleveraging process that has adverse implications for the real economy.

4.8 Sensitivity: Narrative restrictions for shock identification

We have investigated the sensitivity of our results to imposing narrative restrictions for shock identification, extending the proposed identification approach by Rubio-Ramirez et al (2018) to our regime switching model. We combine the narrative restrictions with sign restrictions and compare the results with our findings when using standard sign restrictions presented above. We generally find that our results are rather robust when using narrative restrictions. It is noteworthy, that when using narrative restrictions we need fewer sign restrictions than with the standard sign restriction approach. The results for the RS-VAR including leverage of GSIBs are presented in Appendix C.

Figure 12: Counterfactual probability of staying in the financial constraint regime, GSIBs



Note: Black line: median time-varying probability of staying in financial constraint regime, black line: upper error bound, red line: lower bound; left panel: actual, right pane: counterfactual

5 Conclusions

We conduct a novel empirical analysis on the role of leverage of financial institutions for the transmission of financial shocks to the macroeconomy. To that end we develop an endogenous regime-switching structural vector autoregressive model with time-varying transition probabilities. First, we allow for the transition probabilities to be dependent on the state of the economy, and thereby to be time-varying. Second, we propose new identification schemes for RS-VAR models, extending sign and narrative restrictions to the regime switching model class. To facilitate economic interpretation, we allow the identification restrictions to differ across regimes. One of our contributions is also to highlight a range of identification issues in the context of regime switching models.

Employing this new modelling framework we provide a novel empirical analysis of the role of leverage of financial institutions for the transmission of financial shocks to the macroeconomy. We construct a new monthly market-based measure of leverage of financial institutions as book assets over market equity, building on Adrian and Brunnermeier (2016) by employing financial institution level data. We motivate our focus on leverage with the recent literature on structural macroeconomic models that emphasizes the role of bank balance sheets for the build-up of financial instabilities and the amplification of economic downturns. Furthermore, market leverage encompasses the entire balance sheet of the financial institution and therefore is a broad indicator for signaling financial vulnerability. Finally, we contribute to the empirical literature by, first, presenting empirical evidence that deleveraging can lead to procyclical financial amplification effects with adverse implications for the real economy in the financial constraint regime. Market leverage provides timely information for monitoring the role of financial institutions in that context. Second, we provide evidence for a role of the heterogeneity of financial institutions' leverage for the detrimental real effects of deleveraging of financial institutions, with implications for the probability of persistence of the financial constraint regime. In particular, we analyse the implications of heterogeneity of depository institutions, GSIBs and non-bank financial institutions.

Our empirical evidence indicates the importance of monitoring market-based leverage of financial institutions for financial stability. Two arguments suggest a focus on market leverage: First, Market leverage developments can signal a situation where financial institutions need to deleverage quickly, for instance if debt is used to finance asset growth as for broker-dealers (see Adrian and Shin (2014)) and if financial institutions rely importantly on short-term funding. More generally, we argue that it is important to monitor market leverage for a range of reasons including the effect on financial institutions for coping with liquidity shocks and on funding costs (there is evidence that the cost advantage is larger for banks that are thinly capitalized) relevant from a financial stability perspective and related to the liquidity regulations introduced in Basel III. Second, market values of equity are more informative about financial institutions' losses compared to book values. Book equity values might not be a timely predictor of bank health, since book values incorporate information on losses with a delay, financial institutions have time to adjust their book leverage in order to avoid hitting the regulatory limit. Financial institutions might be more fragile than their book leverage levels make them appear. Furthermore, market capitalization of a bank is a reflection of the market value of the equity holders' stake, and hence an assessment by market participants of the creditworthiness of the bank as a borrower and the present value of the stream of cash flows that derive from the bank's business activities. Low market-to-book ratios suggest that the assessment of market participants is that banks are more leveraged than their books suggest (see also Adrian et al. (2018)). We highlight the role of the financial fragility implied by market leverage for the transmission of financial shocks in our empirical model, that has been pointed to in an estimated model framework (see Begenau et al. (2021)).

It appears that so far the information content of market equity about book losses has been mostly highlighted in the accounting literature, indicating that banks have flexibility in accounting for losses. One might argue that this flexibility in accounting for losses is even more prominent for non-bank financial institutions, highlighting the relevance of our analysis of heterogeneity of financial institutions for regulatory considerations.

Our results on the role of market-based leverage raise the question how to lower the related financial vulnerability. Begenau et al. (2021), for instance, argue that stricter accounting rules (i.e. faster loan loss recognition) could achieve lower financial fragility and thus mitigate the impact of shocks. Aramonte et al. (2021) develop a conceptual framework that emphasises the central role of leverage fluctuations for the propagation of systemic risk due to non-bank financial institutions. We plan to explore the role of non-bank financial institutions further in future research given the heterogeneity of non-bank financial institutions and in light of the current discussion about the appropriate design of non-bank financial institution regulations.

References

- Adrian, T., N. Boyarchenko, and D. Giannone (2019). Vulnerable growth. American Economic Review (109(4)), 1263–89.
- Adrian, T. and M. K. Brunnermeier (2016). CoVar. *The American Economic Review* 106(7), 1705.
- Adrian, T., J. Kiff, and H. S. Shin (2011). Financial intermediary balance sheet management. Annual Reviews of Financial Economics 3, 289–307.
- Adrian, T., J. Kiff, and H. S. Shin (2018). Liquidity, leverage and regulation 10 year after the global financial crisis. *Annual Reviews of Financial Economics* 10, 1–24.
- Adrian, T. and H. S. Shin (2014). Procyclical leverage and value-at-risk. The Review of Financial Studies 27(2), 373–403.
- Aikman, D., A. Lehnert, N. Liang, and M. Modugno (2020). Credit, financial conditions, and monetary policy transmission. *International Journal of Central Banking, International Journal of Central Banking* (16, 3), 141–179.
- Antolín-Díaz, J. and J. F. Rubio-Ramírez (2018). Narrative Sign Restrictions for SVARs. *The American Economic Review* (108, 10), 2802–2829.
- Aramonte, S., A. Schrimpf, and H. S. Shin (2021). Non-bank financial intermediaries and financial stability. *BIS Working Paper* (972, October 2021).
- Arias, J. E., J. F. Rubio-Ramírez, and D. F. Waggoner (2018). Inference based on structural vector autoregressions identified with sign and zero restrictions: Theory and applications. *Econometrica* (86,2), 685–720.
- Aruoba, B., M. Mlikota, F. Schorfheide, and S. Villalvazo (2021). SVARs with occasionallybinding constraints. *Journal of Econometrics*.
- Barnichon, R., C. Matthes, and A. Ziegenbein (2019). Are the effects of financial market disruptions big or small? *Review of Economics and Statistics*, 1–39.

- Bazzi, M., F. Blasques, S. J. Koopman, and A. Lucas (2017). Time-varying transition probabilities for markov regime switching models. *Journal of Time Series Analysis 38*(3), 458–478.
- Begenau, J., S. Bigio, J. Majerovitz, and M. Vieyra (2021). A q-theory of banks. manuscript, Stanford.
- Benigno, G., A. T. Foerster, C. Otrok, and A. Rebucci (2020). Estimating macroeconomic models of financial crises: An endogenous regime-switching approach. *NBER Working Paper* (w26935).
- Blattner, L., L. Farinha, and F. Rebelo (2022). When losses turn into loans: The costs of weak banks. *The American Economic Review*. forthcoming.
- Bognanni, M. and E. Herbst (2018). A sequential monte carlo approach to inference in multiple-equation markov-switching models. *Journal of Applied Econometrics* 33(1), 126– 140.
- Boissay, F., F. Collard, and F. Smets (2016). Booms and banking crises. *Journal of Political Economy* 124(2), 489–538.
- Boyarchenko, N., D. Giannone, and A. Kovner (2020, November). Bank capital and real gdp growth. *Federal Reserve Bank of New York Staff Reports 950*.
- Brunnermeier, M., D. Palia, K. Sastry, and C. Sims (2019). Feedbacks: Financial markets and economic activity. Technical report, Princeton University, Department of Economics, Center for Economic Policy Studies.
- Brunnermeier, M. K. and Y. Sannikov (2014). A macroeconomic model with a financial sector. American Economic Review 104(2), 379–421.
- Chang, Y., Y. Choi, and J. Y. Park (2017). A new approach to model regime switching. *Journal* of *Econometrics* 196(1), 127–143.
- Chang, Y., J. Maih, and F. Tan (2018). State space models with endogenous regime switching. Technical report, Center for Applied Economics and Policy Research, Department of Economics, Indiana University Bloomington.

- Chauvet, M. (1998). An econometric characterization of business cycle dynamics with factor structure and regime switching. *International economic review*, 969–996.
- Chavleishvili, S., R. Engle, S. Fahr, M. Kremer, S. Manganelli, and B. Schwaab (2021). The risk management approach to macro-prudential policy. Technical report, ECB Working Paper Series, No. 2565.
- Diebold, F., J.-H. Lee, and G. Weinbach (1994). Regime switching with time-varying transition probabilities. In C. Hargreaves (Ed.), *Nonstationary Time Series Analysis and Cointegration*, Advanced Texts in Econometrics, pp. 283–302. Oxford University Press.
- Durdu, C. B. and M. Zhong (2019). Understanding bank and nonbank credit cycles: A structural exploration. Technical report, Board of Governors of the Federal Reserve System (US).
- Durham, G., J. Geweke, et al. (2014). Adaptive sequential posterior simulators for massively parallel computing environments. *Advances in Econometrics* 34, 1–44.
- Ferrante, F. (2019). Risky lending, bank leverage and unconventional monetary policy. *Journal* of *Monetary Economics* 101, 100–127.
- Filardo, A. J. (1994). Business-cycle phases and their transitional dynamics. Journal of Business & Economic Statistics 12(3), 299–308.
- Frühwirth-Schnatter, S. (2006). *Finite mixture and Markov switching models*. Springer Science & Business Media.
- Gambacorta, L. and S. Karmakar (2018). Leverage and risk-weighted capital requirements. *International Journal of Central Banking*, 153–191. December 2018.
- Gertler, M. and S. Gilchrist (2018). What happened: Financial factors in the Great recession. *Journal of Economic Perspectives* 32(3), 3–30.
- Gertler, M. and P. Karadi (2011). A model of unconventional monetary policy. Journal of Monetary Economics 58(1), 17–34.

- Gertler, M. and P. Karadi (2013). QE 1 vs. 2 vs. 3...: A framework for analyzing large-scale asset purchases as a monetary policy tool. *International Journal of Central Banking 9*(1), 5–53.
- Gertler, M. and N. Kiyotaki (2010). Financial intermediation and credit policy in business cycle analysis. In *Handbook of Monetary Economics*, Volume 3, pp. 547–599. Elsevier.
- Gertler, M. and N. Kiyotaki (2015). Banking, liquidity, and bank runs in an infinite horizon economy. *American Economic Review* 105(7), 2011–43.
- Gertler, M., N. Kiyotaki, and A. Prestipino (2016). Wholesale banking and bank runs in macroeconomic modeling of financial crises. In *Handbook of Macroeconomics*, Volume 2, pp. 1345–1425. Elsevier.
- Gertler, M., N. Kiyotaki, and A. Prestipino (2020). A macroeconomic model with financial panics. *Review of Economic Studies* 87(2), 240–288.
- Hamilton, J. D. (1989). A new approach to the economic analysis of nonstationary time series and the business cycle. *Econometrica* 57(2), 357–384.
- Hayashi, F. and J. Koeda (2019). Exiting from quantitative easing. *Quantitative Economics* 10(3), 1069–1107.
- He, Z. and A. Krishnamurthy (2019). A macroeconomic framework for quantifying systemic risk. *American Economic Journal: Macroeconomics* 11(4), 1–37.
- Herbst, E. and F. Schorfheide (2014). Sequential Monte Carlo sampling for DSGE models. Journal of Applied Econometrics 29(7), 1073–1098.
- Hubrich, K., A. D'Agostino, M. Cervená, M. Ciccarelli, P. Guarda, M. Haavio, P. Jeanfils,C. Mendicino, E. Ortega, M. T. Valderrama, et al. (2013). Financial shocks and the macroeconomy: heterogeneity and non-linearities. Technical report, ECB Occasional Paper.
- Hubrich, K. and R. J. Tetlow (2015). Financial stress and economic dynamics: The transmission of crises. *Journal of Monetary Economics* 70, 100–115.

- IMF (2009). World Economic Outlook: Sustaining the Recovery, October 2009. International Monetary Fund.
- Jordà, Ô., M. Schularick, and A. M. Taylor (2013). When credit bites back. *Journal of Money*, *Credit and Banking* 45(s2), 3–28.
- Kim, C.-J. (2004). Markov-switching models with endogenous explanatory variables. *Journal* of *Econometrics* 122(1), 127–136.
- Kim, C.-J. and C. R. Nelson (1999). *State-space models with regime switching: classical and Gibbs-sampling approaches with applications*, Volume 1. The MIT press.
- Kim, C.-J., J. Piger, and R. Startz (2008). Estimation of markov regime-switching regression models with endogenous switching. *Journal of Econometrics* 143(2), 263–273.
- Kiyotaki, N. and J. Moore (1997). Credit cycles. *Journal of Political Economy* 105(2), 211–248.
- Mavroeidis, S. (2021). Identification at the zero lower bound. Econometrica.
- Mendoza, E. G. (2010). Sudden stops, financial crises, and leverage. American Economic Review 100(5), 1941–66.
- Paul, P. (2020). A macroeconomic model with occasional financial crises. Journal of Economic Dynamics and Control 112, 103830.
- Rigobon, R. (2003). Identification through heteroskedasticity. *Review of Economics and Statistics* 85(4), 777–792.
- Schularick, M. and A. M. Taylor (2012). Credit booms gone bust: Monetary policy, leverage cycles, and financial crises, 1870-2008. *American Economic Review* 102(2), 1029–61.
- Sims, C. A., D. F. Waggoner, and T. Zha (2008). Methods for inference in large multipleequation markov-switching models. *Journal of Econometrics* 146(2), 255–274.
- Sims, C. A. and T. Zha (1998). Bayesian methods for dynamic multivariate models. *International Economic Review*, 949–968.

- Sims, C. A. and T. Zha (2006). Were there regime switches in us monetary policy? *American Economic Review 96*(1), 54–81.
- Waggoner, D. F., H. Wu, and T. Zha (2016). Striated metropolis-hastings sampler for highdimensional models. *Journal of Econometrics* 192(2), 406–420.
- Waggoner, D. F. and T. Zha (2003). Likelihood preserving normalization in multiple equation models. *Journal of Econometrics* 114(2), 329–347.

A Proof of Proposition 1

The following lemma is key to proving Proposition 1. In both the statement of the lemma and its proof, it will always be assumed that $1 \le k \le r$ and $1 \le m \le s$.

Lemma 1 Let A_1, \dots, A_r be invertible $n \times n$ matrices and let D_1, \dots, D_s be distinct $n \times n$ diagonal matrices such that the diagonal elements of each D_m are distinct and $\sum_{m=1}^s D_m = I_n$. If $\tilde{A}_1, \dots, \tilde{A}_r$ are $n \times n$ matrices, $\tilde{D}_1, \dots, \tilde{D}_s$ are $n \times n$ diagonal matrices such that $\sum_{m=1}^s \tilde{D}_m = I_n$, $\pi(k,m)$ is a function that is a permutation of $\{1, \dots, s\}$ for each k, and $A'_k D_m A_k = \tilde{A}'_k \tilde{D}_{\pi(k,m)} \tilde{A}_k$, then $\pi(k,m)$ is independent of k, $\tilde{A}_k = E_k Q' A_k$ and $\tilde{D}_{\pi(k,m)} = Q' D_m Q$, where Q is a permutation matrix and E_k is a diagonal matrix with plus or minus ones along the diagonal.

Proof. Since $\pi(k, m)$ is a permutation of $\{1, \dots, s\}$ for each *k*,

$$I_n = (A'_k)^{-1} \left(\sum_{m=1}^s A'_k D_m A_k \right) A_k^{-1} = (A'_k)^{-1} \left(\sum_{m=1}^s \tilde{A}'_k \tilde{D}_{\pi(k,m)} \tilde{A}_k \right) A_k^{-1} = (\tilde{A}_k A_k^{-1})' (\tilde{A}_k A_k^{-1}),$$

it follows that $(\tilde{A}_k A_k^{-1})' \equiv \hat{Q}_k$ is an orthogonal matrix and $\tilde{A}_k = \hat{Q}'_k A_k$. Thus,

$$A'_k D_m A_k = \tilde{A}'_k \tilde{D}_{\pi(k,m)} \tilde{A}_k = A'_k \hat{Q}_k \tilde{D}_{\pi(k,m)} \hat{Q}'_k A_k,$$

which implies that $D_m = \hat{Q}_k \tilde{D}_{\pi(k,m)} \hat{Q}'_k$. Because the eigenvalues of a symmetric matrix are unique up to an ordering, the D_m are distinct, and the diagonal elements of each D_m are distinct, the permutation $\pi(k,m)$ does not depend on k. If Q is the column permutation matrix associated with π , then for each k there exists a diagonal matrix E_k , with plus or minus ones along the diagonal, such that $\hat{Q} = QE_k$.

Proof of Proposition 1. Throughout this proof, it will be assumed that $1 \le k \le h_c$, $1 \le m \le h_v$, and $1 \le t \le T$. Let $M(k) = A_0^{-1}(k)A_+(k)$ and $V(k,m) = A_0^{-1}(k)\Xi^{-2}(m)A_0^{-1}(k)'$. For almost all $A_0(k)$, $A_+(k)$ and $\Xi(m)$, the M(k) are distinct, the V(k,m) are distinct, the $\Xi(m)$ are distinct, and the diagonal elements of each $\Xi(m)$ are distinct. So, we can assume that the M(k) are distinct, the V(k,m) are distinct, the $\Xi(m)$ are distinct, and the diagonal elements of each $\Xi(m)$ are distinct.

The distribution of y_t , conditional on x_t , is a mixture of $h = h_c h_v$ normal distributions with means $M(k)x_t$, variances V(k,m), and weights $p((s_t^c, s_t^v) = (k,m)|x_t)$. For a mixture of *distinct*

normal distributions with positive weights, the means and variances of the normal distributions in the mixture and their weights are uniquely determined by the distribution of the mixture. If some of the weights were zero, then the normal distributions corresponding to those zero weights would not be determined, though the weights themselves would be. Because the V(k,m) are distinct, the conditional distribution of y_t is a mixture of h distinct normal normal distributions.

If the conditional probabilities, $p((s_t^c, s_t^v) = (k, m)|x_t)$, were zero for all x_t , then the unconditional probabilities, $p((s_t^c, s_t^v) = (k, m))$, would also be zero. So by the hypotheses of Proposition 1, $p((s_t^c, s_t^v) = (k, m)|x_t)$ cannot be identically zero for all t. Because the weight associated with each (k, m) is non-zero for some t, this implies that the normal distributions and their weights are uniquely determined by the conditional distributions of the y_t .

If the system $\tilde{A}_0(\tilde{s}_t^c)y_t = \tilde{A}_+(\tilde{s}_t^c)x_t + \tilde{\Xi}^{-1}(\tilde{s}_t^v)\varepsilon_t$ generates observationally equivalent data as the system given by Equation (1), then there must exist functions $\pi_c(k, m, t) \in \{1, \dots, h_c\}$ and $\pi_v(k, m, t) \in \{1, \dots, h_v\}$ such that

$$M(k)x_t = \tilde{A}_0^{-1}(\pi_c(k, m, t))\tilde{A}_+(\pi_c(k, m, t))x_t$$
(8)

$$V(k,m) = \left(\tilde{A}_0^{-1}(\pi_c(k,m,t))\tilde{\Xi}^{-2}(\pi_v(k,m,t))\tilde{A}_0^{-1}(\pi_c(k,m,t))'\right)^{-1}$$
(9)

By the hypotheses of Proposition 1, the predetermined data span all of \mathbb{R}^m , so Equation (8) implies that

$$M(k) = \tilde{A}_0^{-1}(\pi_c(k, m, t))\tilde{A}_+(\pi_c(k, m, t))$$
(10)

Since the left hand sides of Equations (9) and (10) do not depend on *t*, neither can the right hand side. Since the both the M(k) and the V(k,m) are distinct, this implies that neither π_c nor π_v depend on *t*. Since the left hand side of Equation (10) does not depend on *m* and the M(k) are distinct, π_c does not depend on *m*. So we can write $\pi_c(k,m,t) = \pi_c(k), \pi_v(k,m,t) = \pi(k,m)$ and Equations (9) becomes

$$V(k,m) = \left(\tilde{A}_0^{-1}(\pi_c(k))\tilde{\Xi}^{-2}(\pi_v(k,m))\tilde{A}_0^{-1}(\pi_c(k))'\right)^{-1}$$
(11)

Because the V(k,m) are distinct, it must be the case that $\pi_v(k,m)$ is a permutation of $\{1, \dots, h_v\}$,

for each k. We can now apply Lemma 1 with $r = h_c$, $s = h_v$, and $\pi(k,m) = \pi_v(k,m)$. When the system is normalized so that $\sum_{m=1}^{h_v} \Xi^2(m) = I_n$, we take $A_k = A_0(k)$, $\tilde{A}_k = \tilde{A}_0(\pi_c(k))$, $D_m = \Xi^2(m)$, and $\tilde{D}_m = \tilde{\Xi}^2(m)$. When the system is normalized by $\sum_{m=1}^{h_v} \Xi^{-2}(m) = I_n$, we take $A_k = A_0^{-1}(k)'$, $\tilde{A}_k = \tilde{A}_0^{-1}(\pi_c(k))'$, $D_m = \Xi^{-2}(m)$, and $\tilde{D}_m = \tilde{\Xi}^{-2}(m)$. In either case, we obtain that $\pi_v(k,m)$ does not depend on k and can be written as $\pi_v(k,m) = \pi_v(m)$ and there exists a permutation matrix Q and diagonal matrices E_k , with plus or minus ones along the diagonal, such that $\tilde{A}_0(\pi_c(k)) = E_k Q' A_0(k)$ and $\Xi(\pi_v(m)) = Q' \Xi(m)Q$. Thus, Equation (10) implies that $\tilde{A}_+(\pi_c(k)) = E_k Q' A_+(k)$. So, under the hypotheses of Proposition 1, for almost all $A_0(\cdot)$, $A_+(\cdot)$, and $\Xi(\cdot)$, the system given by Equation (1) is identified up to the sign and ordering of the equations and the ordering of the regimes.

B The Likelihood, Posterior, Filtered and Smoothed Probabilities

In this appendix we derive the formulas for computing the the likelihood, the posterior, and filtered and smoothed probabilites. The likelihood is

$$p(Y_T|Z_T, \theta) = \frac{p(Y_T, Z_T|\theta)}{p(Z_T|\theta)}$$
(12)

$$=\frac{\prod_{t=1}^{T} p(y_t, z_t | Y_{t-1}, Z_{t-1}, \theta)}{p(Z_T | \theta)}$$
(13)

$$=\frac{\prod_{t=1}^{T}\sum_{s_{t}=1}^{h}p(s_{t}, y_{t}, z_{t}|Y_{t-1}, Z_{t-1}, \theta)}{p(Z_{T}|\theta)}$$
(14)

$$=\frac{\prod_{t=1}^{T}\sum_{s_{t}=1}^{h}p(y_{t}|s_{t},Y_{t-1},Z_{t},\theta)p(z_{t}|s_{t},Y_{t-1},Z_{t-1},\theta)p(s_{t}|Y_{t-1},Z_{t-1},\theta)}{p(Z_{T}|\theta)}$$
(15)

$$=\frac{\prod_{t=1}^{T} p(z_t|Y_{t-1}, Z_{t-1})}{p(Z_T|\theta)} \prod_{t=1}^{T} \sum_{s_t=1}^{h} p(y_t|s_t, Y_{t-1}, Z_t, \theta) p(s_t|Y_{t-1}, Z_{t-1}, \theta),$$
(16)

where Equations (12), (13), and (15) follow from Bayes' rule, Equation (14) is obtained by integrating out s_t , and Equation (16) follows from Assumption 1 and a rearrangement of terms. Note that Assumption 1 does not, in general, imply that $p(Z_t|\theta) = p(Z_t)$. If $p(\theta)$ is the prior, then the posterior is

$$P(\boldsymbol{\theta}|\boldsymbol{Y}_T, \boldsymbol{Z}_T) = \frac{p(\boldsymbol{Y}_T, \boldsymbol{Z}_T, \boldsymbol{\theta})}{p(\boldsymbol{Y}_T, \boldsymbol{Z}_T)}$$
(17)

$$=\frac{p(Y_T|Z_T,\theta)p(Z_T|\theta)p(\theta)}{p(Y_T,Z_T)}$$
(18)

$$=\frac{\prod_{t=1}^{T} p(z_t|Y_{t-1}, Z_{t-1})}{p(Y_T, Z_T)} p(\theta) \prod_{t=1}^{T} \sum_{s_t=1}^{h} p(y_t|s_t, Y_{t-1}, Z_t, \theta) p(s_t|Y_{t-1}, Z_{t-1}, \theta), \quad (19)$$

where Equations (17) and (18) follow from Bayes rule and Equation (19) follows by substituting the expression for the likelihood and canceling and rearranging terms. So, the posterior is proportional to

$$p(\theta) \prod_{t=1}^{T} \sum_{s_t=1}^{h} p(y_t|s_t, Y_{t-1}, Z_t, \theta) p(s_t|Y_{t-1}, Z_{t-1}, \theta).$$

The recursive formulas for the the Hamilton filter are derived next.

$$p(s_{t+1}|Y_t, Z_t, \theta) = \sum_{s_t=1}^h p(s_{t+1}, s_t | Y_t, Z_t, \theta)$$
(20)

$$= \sum_{s_t=1}^{h} p(s_{t+1}|s_t, Y_t, Z_t, \theta) p(s_t|Y_t, Z_t, \theta),$$
(21)

where Equation (20) is obtained by integrating out s_t and Equation (21) follows from Bayes' rule.

$$p(s_t|Y_1^t, Z_1^t, \theta) = \frac{p(s_t, y_t, z_t|Y_1^{t-1}, Z_1^{t-1}, \theta)}{p(y_t, z_t|Y_1^{t-1}, Z_1^{t-1}, \theta)}$$
(22)

$$=\frac{p(s_t, y_t, z_t | Y_1^{t-1}, Z_1^{t-1}, \theta)}{\sum_{s_t=1}^h p(s_t, y_t, z_t | Y_1^{t-1}, Z_1^{t-1}, \theta)}$$
(23)

$$=\frac{p(y_t|s_t, Y_1^{t-1}, Z_1^t, \theta)p(z_t|s_t, Y_1^{t-1}, Z_1^{t-1}, \theta)p(s_t|Y_1^{t-1}, Z_1^{t-1}, \theta)}{\sum_{s_t=1}^h p(y_t|s_t, Y_1^{t-1}, Z_1^t, \theta)p(z_t|s_t, Y_1^{t-1}, Z_1^{t-1}, \theta)p(s_t|Y_1^{t-1}, Z_1^{t-1}, \theta)}$$
(24)

$$=\frac{p(z_t|Y_1^{t-1},Z_1^{t-1})p(y_t|s_t,Y_1^{t-1},Z_1^{t},\theta)p(s_t|Y_1^{t-1},Z_1^{t-1},\theta)}{p(z_t|Y_1^{t-1},Z_1^{t-1})\sum_{s_t=1}^h p(y_t|s_t,Y_1^{t-1},Z_1^{t},\theta)p(s_t|Y_1^{t-1},Z_1^{t-1},\theta)}$$
(25)

$$=\frac{p(y_t|s_t, Y_1^{t-1}, Z_1^{t}, \theta)p(s_t|Y_1^{t-1}, Z_1^{t-1}, \theta)}{\sum_{s_t=1}^h p(y_t|s_t, Y_1^{t-1}, Z_1^{t}, \theta)p(s_t|Y_1^{t-1}, Z_1^{t-1}, \theta)},$$
(26)

where Equations (22) and (24) follow from Bayes' rule, Equation (23) is obtained by integrating out s_t in the denominator, Equation (25) follows from Assumption 1 and rearranging terms, and Equation (26) follows by canceling terms. Before deriving the formulas for the smoothed probabilities, we must first develop more notation. For $1 \le \tau \le t \le T$, let S_{τ}^t denote $(s_{\tau}, s_{\tau+1}, \dots, s_t)$. We also need the following assumption:

Assumption 2

$$p(s_t|Y_{t-1}, Z_{t-1}, S_{t-1}, \theta) = p(s_t|Y_{t-1}, Z_{t-1}, s_{t-1}, \theta)$$
(27)

$$p(y_t|Y_{t-1}, Z_{t-1}, S_{t-1}, \theta) = p(y_t|Y_{t-1}, Z_{t-1}, s_{t-1}, \theta)$$
(28)

For the models discussed in this paper, Assumption 2 holds. It is possible to weaken Assumption 2 by replacing s_{t-1} on the right hand sides of Equations (27) and (28) with S_{t-k}^{t-1} , for some fixed value of $k \ge 1$.

It follows from Bayes' rule and Assumptions 1 and 2 Equation (27) that for $1 \le \tau \le t \le T$

$$p(y_t, z_t | S_{\tau}^t, Y_{t-1}, Z_{t-1}, \theta) = p(y_t | s_t, Y_{t-1}, Z_t, \theta) p(z_t | Y_{t-1}, Z_{t-1}).$$
⁽²⁹⁾

So, $1 \le \tau_0 \le \tau_1 \le t \le T$

$$p(y_t, z_t | S_{\tau_0}^t, Y_{t-1}, Z_{t-1}, \theta) = p(y_t, z_t | S_{\tau_1}^t, Y_{t-1}, Z_{t-1}, \theta).$$
(30)

For $1 \le t+1 < \tau \le T$,

$$p(s_t|S_{t+1}^{\tau}, Y_{\tau}, Z_{\tau}, \theta) = \frac{p(s_t, s_{\tau}, y_{\tau}, z_{\tau}|S_{t+1}^{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta)}{p(s_{\tau}, y_{\tau}, z_{\tau}|S_{t+1}^{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta)}$$
(31)

$$=\frac{p(y_{\tau}, z_{\tau}|S_{t}^{\tau}, Y_{\tau-1}, Z_{\tau-1}, \theta)p(s_{\tau}|S_{t}^{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta)p(s_{t}|S_{t+1}^{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta)}{p(y_{\tau}, z_{\tau}|S_{t+1}^{\tau}, Y_{\tau-1}, Z_{\tau-1}, \theta)p(s_{\tau}|S_{t+1}^{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta)}$$
(32)

$$=\frac{p(s_{\tau}|s_{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta)p(s_t|S_{t+1}^{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta)}{p(s_{\tau}|s_{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta)}$$
(33)

$$= p(s_t | S_{t+1}^{\tau-1}, Y_{\tau-1}, Z_{\tau-1}, \theta),$$
(34)

where Equations (31) and (32) follow from Bayes' rule, Equation (33) follows from Equation (30), cancellation, and Assumption 2 Equation (28), and Equation (34) follows from cancellation. Thus, by a recursive argument,

$$p(s_t|S_{t+1}^T, Y_T, Z_T, \theta) = p(s_t|s_{t+1}, Y_{t+1}, Z_{t+1}, \theta).$$
(35)

But

$$p(s_t|s_{t+1}, Y_{t+1}, Z_{t+1}, \theta) = \frac{p(s_t, s_{t+1}, y_{t+1}, Z_{t+1}|Y_t, Z_t, \theta)}{p(s_{t+1}, y_{t+1}, Z_{t+1}|Y_t, Z_t, \theta)}$$
(36)

$$=\frac{p(y_{t+1}, z_{t+1}|s_{t+1}, s_t, Y_t, Z_t, \theta)p(s_{t+1}|s_t, Y_t, Z_t, \theta)p(s_t|Y_t, Z_t, \theta)}{p(y_{t+1}, z_{t+1}|s_{t+1}, Y_t, Z_t, \theta)p(s_{t+1}|Y_t, Z_t, \theta)}$$
(37)

$$=\frac{p(s_{t+1}|s_t, Y_t, Z_t, \theta)p(s_t|Y_t, Z_t, \theta)}{p(s_{t+1}|Y_t, Z_t, \theta)},$$
(38)

where Equations (36) and (37) follow from Bayes' rule, and Equation (38) follows from Equation (30) and cancellation. Thus

$$p(s_t|S_{t+1}^T, Y_T, Z_T, \theta) = \frac{p(s_{t+1}|s_t, Y_t, Z_t, \theta)p(s_t|Y_t, Z_t, \theta)}{p(s_{t+1}|Y_t, Z_t, \theta)}.$$
(39)

Since

$$p(s_t|Y_T, Z_T, \theta) = \sum_{S_{t+1}^T} p(s_t, S_{t+1}^T | Y_T, Z_T, \theta)$$
(40)

$$= \sum_{S_{t+1}^T} p(s_t | S_{t+1}^T, Y_T, Z_T, \theta) p(S_{t+1}^T | Y_T, Z_T, \theta)$$
(41)

$$= p(s_t|Y_t, Z_t, \theta) \sum_{s_{t+1}=1}^h \left(\frac{p(s_{t+1}|s_t, Y_t, Z_t, \theta) p(s_{t+1}|Y_T, Z_T, \theta)}{p(s_{t+1}|Y_t, Z_t, \theta)} \times \sum_{\substack{S_{t+2}^T}} p(S_{t+2}^T|s_{t+1}, Y_T, Z_T, \theta) \right)$$
(42)

$$= p(s_t|Y_t, Z_t, \theta) \sum_{s_{t+1}=1}^{h} \frac{p(s_{t+1}|s_t, Y_t, Z_t, \theta) p(s_{t+1}|Y_T, Z_T, \theta)}{p(s_{t+1}|Y_t, Z_t, \theta)},$$
(43)

where Equation (40) is obtained by integrating out S_{t+1}^T , Equation (41) follows from Bayes' rule, Equation (42) follows from Equation (39) and rearranging terms, and Equation (43) follows because $p(S_{t+2}^T|s_{t+1}, Y_T, Z_T, \theta)$ is a density and integrates (sums) to one.

C Impulse responses with narrative restrictions

Figure A.1: Impulse Responses to a financial shock in the financial constraint regime, narrative sign restrictions, GSIBs



Note: Impulse Responses to financial shock, narrative sign restrictions, (conditional on staying in) the financial constraint regime; red line: median response, blue lines: lower and upper bound of the 68 percent error bounds; financial shock: 1 std shock to financial conditions index; identification with contemporaneous sign restrictions, median and 68% error bands, output growth (IP), core inflation (CPI), 2-y Treasury yield, market leverage, Financial conditions index (Chicago Fed)

Figure A.1 presents the results for the financial constraint regime for a model with leverage of GSIBs and identification with narrative sign restrictions, where the shock is identified that explains most of the variation in output growth during GFC.

Responses to a financial shock (median) show a protracted negative output response. Market leverage initially increases, then declines due to deleveraging. We arrive at the same conclusion as in our baseline specification for the GSIBs: Deleveraging can lead to amplification effects with adverse implications for the real economy. In normal times, the responses to a financial shock (median) indicate small, nonpersistent negative output response and insignificant market leverage (Figure A.2) as with the standard sign restriction approach.

Figure A.2: Impulse Responses to a financial shock in normal regime, narrative sign restrictions, GSIBs



Note: Impulse Responses to financial shock, narrative sign restrictions, (conditional on staying in) the financial constraint regime; red line: median response, blue lines: lower and upper bound of the 68 percent error bounds; financial shock: 1 std shock to financial conditions index; identification with contemporaneous sign restrictions, median and 68% error bands, output growth (IP), core inflation (CPI), 2-y Treasury yield, market leverage, Financial conditions index (Chicago Fed)

D Time-varying Probabilities with error bands

Figure A.3: Time-varying probability of staying in low interest rate regime with 68 % error bands, conditional of being in that regime, GSIBs

